Copyright 2015 Chandrasekaran Jayaraman

SHOULDER PAIN IN MANUAL WHEELCHAIR USERS: TOWARDS A MULTI-DISCIPLINARY SOLUTION FOR A MULTI-FACETED PROBLEM

ΒY

CHANDRASEKARAN JAYARAMAN

DISSERTATION

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Systems and Entrepreneurial Engineering in the Graduate College of the University of Illinois at Urbana-Champaign, 2015

Urbana, Illinois

Doctoral Committee:

Associate Professor Carolyn Beck, Chair & Director of Research Associate Professor Jacob Sosnoff, Co-Director of Research Assistant Professor Ian Rice Professor Alex Kirlik Associate Professor Ramavarapu Sreenivas

ABSTRACT

It is estimated that there are over 2 million manual wheelchair users in the United States. Up to 70% of manual wheelchair users report upper limb pain, which is mainly manifested in the shoulder and wrist. Shoulder pain in wheelchair users is linked to difficulty performing activities of daily living, decreased physical activity and decreased quality of life[.]

The main focus of this dissertation is to identify biomarkers from wheelchair propulsion data that are potentially related to shoulder pain in manual wheelchair users. Three biomarkers that distinguish between manual wheelchair users with and without shoulder pain are identified. The acceptability of the identified biomarkers are subjected to hypothesis testing using data collected from a sample of 30 experienced adult manual wheelchair users with and without shoulder pain. The results and their implications will be discussed. In this dissertation we will also discuss the interpretation and the physical significance of each of the results, a summary of limitations for the approaches adopted, and suggestions on the future course of research to address these limitations.

While the past two decades of research on shoulder pain and wheelchair propulsion has led to the development of important clinical guidelines, it has failed to identify specific biomarkers that may be related to shoulder pain in manual wheelchair users. This could be in part due to employing a binary approach by focusing on just (1) the pure bio-mechanical

ii

aspects, and (2) wheelchair design aspects (ergonomics). The originality of this dissertation is in the adoption of a multidisciplinary approach. Methodologies integrating theories and analyses from fields related to human movement science such as human motor control theory, non-linear dynamics and human factors (occupational ergonomics) are adopted to identify potential biomarkers that relate to shoulder pain in manual wheelchair users.

This dissertation concludes with preliminary results from a prototype wearable device, custom developed for manual wheelchair users. Wheelchair propulsion data obtained from the device will be benchmarked with data from the currently available technologies for tracking manual wheelchair propulsion (SMARTWheel and motion capture). This dissertation also proposes a framework for incorporating the research findings into the custom developed wearable technology for home-based rehabilitation training purposes.

iii

To my grandmother, mother, & brother for their continued support all along, And Shri. Gopalakrishnan for being a constant source of moral guidance. आचार्यात् पादमादत्ते

आचार्यात् पादमादत्ते पादं शिप्यः स्वमेधया । सत्रह्मचारिभ्यः पादं पादं कालक्रमेण च ॥

Sanskrit verse, Rig Veda

Translates as-

Knowledge acquisition

one fourth from the teacher, one fourth from own intelligence, one fourth from fellow classmates, and one fourth only with time.

ACKNOWLEDGEMENTS

I express my gratitude to Prof. Carolyn Beck (Department of ISE, UIUC) for her extremely valuable advice, support and mentorship throughout my period as a graduate student. Dr. Beck has been a great source of inspiration and motivation during the up's and downs of my graduate life at UIUC.

I express by sincere gratitude to Prof. Sosnoff (Department of KCH, UIUC) for having supported me and mentored me throughout my graduate program. His honest feedback, patience and enthusiasm towards brainstorming research ideas, selflessness with time for his students and stimulating discussions have been the main propellant towards this achievement. Above all, I believe that my association with both, Prof. Sosnoff and Prof. Carolyn has made me a more kind towards world.

I thank Prof. Ian Rice (Department of KCH, UIUC) for his constant insight, support and guidance during my PhD research.

I thank my doctoral committee members, Prof. Sreenivas and Prof. Kirlik for their valuable insights and constructive feedback on the dissertation research. They have been extremely generous with the time they spent interacting with me during the final phases of the dissertation. Their insights on this dissertation have inspired me a lot and will stay to influence my future research thinking.

I thank the Department of ISE, Department of KCH, and the Graduate college UIUC, for funding parts of my graduate studies. This research work was also supported in parts by the National Institute of Health (#1R21HD066129-01A1). I specifically, want to thank the Chittenden family

vi

for offering me the Chittenden Fellowship (2009-2010) which was the main driving mechanism that facilitated this collaborative research between the Department of ISE and the Department of KCH

I thank the help offered by my colleagues during the data collection for this research work. (Mr. Mike Socie, Ms. Iris, Ms. Yaejin Moon, Dr. Karla Wessels, Ms. Katie White, Ms. Shawna Culp and others). I also thank MCRL members, Ms. Kathleen (Katie), Mr. Jong and Mr & Ms. Wajda for their continued encouragement and support.

I would like to thanks the staff members from both, the department of KCH and department of ISE at UIUC for being very helpful and patient with me while doing all the administration paper work. Specifically I would like to thank Tina, Holly and Julie for their patience in helping me with all the paper work.

I like to thank my friends, Sriram, Ashwin, Sambu, Ismail, Adam, Muda, Meng, Vivek, Sangeetha, Nisha, Bharat, Srinivas, Anshuman, Prateek, Shankar, Mayank, Martin and many more, for the very nice memories we shared during grad school.

Finally and most of all I would like to thank all the participants who took time and volunteered to participate in this study.

TABLE OF CONTENTS

Chapter 1 – Introduction and motivation1
Chapter 2 – Experimental data collection procedures15
Chapter 3 – Shoulder pain and jerk during wheelchair propulsion23
Chapter 4 – Shoulder pain and trunk kinematics in manual
wheelchair propulsion56
Chapter 5 – Relationship between cycle-to-cycle structure in
variability and shoulder pain in manual wheelchair users
Chapter 6 – Prototype and validation of custom wearable
technology for manual wheelchair users114
Chapter 7 – Overall conclusions and future directions
Chapter 8 – Intellectual contributions to wheelchair
propulsion research144

Chapter 1

Introduction and motivation

It is estimated that over 3.6 million Americans use wheelchairs for mobility (LaPlante et al., 2010) with a majority (~90%) using a manual wheelchair (LaPlante et al., 2010). Although wheelchair use has numerous benefits (Hosseini et al., 2012), the repetitive cyclic arm movement required for manual propulsion places a significant demand on the upper extremity, specifically the shoulder (Nichols et al., 1979, Gellman et al., 1988, Curtiset al., 1999, Finley et al., 2004). This increased demand often results in shoulder pain. Indeed up to 70% of manual wheelchair users report shoulder pain (Finley et al., 2004).

Shoulder pain in wheelchair users have been linked to difficulty performing activities of daily living, decreased physical activity and decreased quality of life (Chow et al., 2011). Subsequently, it is imperative to understand the mechanisms that contribute to shoulder pain in manual wheelchair users so that appropriate interventions can be developed to prevent or minimize the effect of shoulder pain on function and thus reduce the risk of long-term upper extremity disability.

Shoulder pain occurring in wheelchair users is a multi-faceted problem (Dyson-Hudson et al., 2004). It has been suggested that upper limb pain is related to a myriads of factors including functional level (Curtis KA et al., 1999), duration of wheelchair use, wheelchair design (van der Woude et al., 2006), body weight (Sinnott et al., 2000, Collinger et al., 2008), propulsion mechanics (Koontz et al., 2002, Mercer et al., 2006) muscle coordination (Burhham et al., 1993, Kotajarvi et al., 2002), age (Fullerton et al., 2003), and gender (Lal S, 1998, Gutierrez et al., 2007). The multifactorial nature of the possible mechanisms and associated variables creates a daunting task for researchers and clinicians.

Given that the association between shoulder pain and manual wheelchair propulsion is multi-faceted in nature, a multi-disciplinary approach to analyze and address this problem is needed. Such multidisciplinary approaches can provide better understanding of the pathology and may lead to new knowledge for better monitoring/tracking/ and prevention of shoulder injury in manual wheelchair users.

Investigations of mechanisms contributing to shoulder pain in wheelchair users have mainly examined biomechanical variables of manual wheelchair propulsion (Boninger et al., 2002, Collinger et al., 2008, Rankin et al., 2011, Richter et al., 2007, 2011, Koontz et al., 20009, 2012, de Groot et al., 2003,2004, Kwarciak et al., 2012) and wheelchair design aspects

(van der Woude et al., 2006, Boninger et al., 2005, Cowan et al., 2009). While the research on shoulder pain and wheelchair propulsion has provided important information and has led to the development of clinical guidelines (Boninger et al., 2002, Collinger et al., 2008, Rankin et al., 2011, 2012, Richter et al., 2007, 2011, Koontz et al., 20009, 2012, de Groot et al., 2003,2004, Kwarciak et al., 2012, Mercer et al., 2006), it has several potential limitations.

First, a multi-disciplinary approach to examine wheelchair propulsion has not been adopted. Second, research has mainly focused on the complete propulsion cycle and the push phase, but much less so on the recovery phase. Third, there has been minimal enquiry on the temporal structure of motor variability in wheelchair propulsion in the context of shoulder pain.

Research objectives of this dissertation

Consequently the main objectives of this dissertation is to implement a multi-disciplinary approach to identify biomarkers from wheelchair propulsion data that could be potentially related to shoulder pain in manual wheelchair users. Methodologies integrating theories and analyses from fields related to human movement science such as human motor control theory, non-linear dynamics and human factors (occupational ergonomics) are adopted to identify potential biomarkers that relate to shoulder pain in

manual wheelchair users. Three biomarkers that distinguish between manual wheelchair users with and without shoulder pain are identified. These three biomarkers are: (1) differences in kinematic jerk during recovery phase between individuals propelling a manual wheelchair with and without shoulder pain (Chapter 3), (2) trunk kinematic differences between individuals propelling a manual wheelchair with and without shoulder pain (Chapter 4) and (3) the structure in cycle-to-cycle variability of wheelchair propulsion variables between individuals propelling a manual wheelchair manual wheelchair with and without shoulder pain (Chapter 5).

The acceptability of the identified biomarkers are evaluated sing hypothesis testing data collected from a sample of 30 experienced adult manual wheelchair users with and without shoulder pain. The results and their implications will be discussed. This dissertation will also discuss the interpretation and the physical significance of each of the results, a summary of limitations for the approaches adopted, and suggestions on the future course of research to address these limitations.

Dissertation organization and chapter associations

This dissertation is organized as follows:

Chapter 2 describes the experimental data collection methodologies. The data derived from the same experimental setup was used to hypotheses test

all three studies reported in Chapters 3, 4 and 5.

Chapter 3 contains the motivation, methodologies, results and discussion for Study 1: Shoulder pain and jerk during wheelchair propulsion.

Chapter 4 reports the motivation, methodologies, results and discussion for Study 2: Shoulder pain and trunk kinematics during wheelchair propulsion. *Chapter 5* contains details about Study 3 which investigates the relationship between the cycle-to-cycle time dependent structure in variability and shoulder pain in manual wheelchair users.

Chapter 6 takes the insights from the three different studies and proposes a framework for the implementation of these metrics through a wearable technology for day-to-day monitoring of wheelchair user propulsion mechanics.

Chapter 7 contains the overall conclusion and suggestions for future direction of research. Chapter 8 contains list of intellectual contribution this dissertation to wheelchair propulsion literature.

Association between chapters (Figure 1.1)

Chapter's 1 and 2 contain the necessary preliminary background materials for Chapters 3, 4 and 5. The Chapters 3, 4 and 5 are not interrelated and can be read in any order. However, it is recommended that to fully appreciate the contents of Chapters 6, 7 and 8, all the previous chapters be read (Figure 1.1).

The multi-disciplinary approach adopted

Contrary to traditional approaches, this dissertation adopts a multidisciplinary approach to identify biomarkers that related to shoulder pain in manual wheelchair users (Figure 1.2). The multi-disciplinary area that this dissertation relies on were not considered in previous in wheelchair propulsion literature. The previous literature focused on just two aspects namely, propulsion biomechanics and wheelchair ergonomics (design aspects), while this dissertation provides a new dimension to this problem integrating approaches from human movement science disciplines like, human motor control theory, non-linear dynamics and human movement ergonomics (occupational ergonomics).

Basic wheelchair propulsion terminologies

Before we elaborate further on each of the biomarkers considered, it is important to define some standard manual wheelchair propulsion terminologies used throughout this dissertation.

A typical manual wheelchair propulsion cycle consists of a push phase (i.e. when the hand is in contact with the hand-rim/wheel) and a recovery phase (when the hand is off the hand-rim/wheel). During the push phase the arms are constrained to follow the hand-rims, while during recovery phase the arms can adopt a variety of different movement patterns (Figure 1.1). Four typical propulsion pattern types have been observed based on the hand

trajectory during the recovery phase of manual wheelchair propulsion. They are a semi-circular (SC) pattern, double loop pattern (DLOP), single loop pattern (SLOP) and an arc pattern (Sanderson et al., 1985, Shimada et al., 1988, Boninger et al., 2002, Richter et al., 2007).

A DLOP pattern is characterized by the hands lifting over the propulsion path and crossing the propulsion path to drop below the hand-rim forming a double loop, while a SC pattern is characterized by the hands dropping below the hand-rim during the recovery phase, the hands rise above the hand rim during the recovery phase for a SLOP pattern and) the hands follows the hand rim closely during the recovery phase for a ARC pattern forming a pumping action (Figure 1.2) (Sanderson et al., 1985, Shimada et al., 1988, Boninger et al., 2002, Richter et al., 2007).

In this dissertation is goes, to capture, motion data (i.e movement kinematics), a motion capture system was used to capture movement kinematics. The procedures and experimental setup for this are detailed in Chapter 2 of this dissertation.

<u>Peak resultant force at hand-rim</u> (Newton (N)): The peak value of the resultant force applied at the hand-rim by the palm to push the wheelchair (Figure 1.4(a, b)).

<u>Contact angle</u> (degrees (θ^{o})): The angle for which the hand is in contact with the hand-rim during the push phase of wheelchair propulsion (i.e. while

pushing the wheelchair (Figure 1.4 (b)).

<u>Inter push time interval between peak resultant force</u> (seconds (sec)): The time interval between two consecutive peak resultant forces at hand-rim (Figure 1.4 (b)).

<u>Push time</u> (seconds (sec)): The time taken from the start to end of a push phase (Figure 1.3 (b)).

<u>Recovery time</u> (seconds (sec)): The time taken from the end of a push phase to the start of the consecutive push phase (Figure 1.4 (b)).

<u>Steady state propulsion</u>: The portion of the wheelchair propulsion trial from the 6th push till the end of the trial (Figure 1.4 (a)). The first 5 pushes are influenced by initial start-up effects in overcoming inertia and research evidence suggests that it usually takes up to five pushes to reach a steady state pushing rate (Koontz et al., 2009).

List of figures



Figure 1.1 Association between chapters



Multi-disciplinary approach

Figure 1.2 Various multi-disciplinary approaches this dissertation adopts to investigate shoulder pain in manual wheelchair users



Figure 1.3 Different recovery pattern types. (a) Double loop (DLOP); (b) Semicircular (SC); (c) Single loop (SLOP) and (d) Arc. The dotted lines denote the push phase and the solid line the recovery phase.



Figure 1.4 Basic terminologies of wheelchair propulsion mechanics used in this dissertation. (a) A plot showing the resultant force at hand-rim for a 3 minute trial with approximately 120 pushes. The steady state portion of the trial is also shown; (b) a magnified view of two sample pushes cycles from the full trial showing all the important terminologies, namely, contact angle, peak resultant force, inter push time interval between peak resultant force, push time and recovery time.

References

- Boninger ML, Koontz AM, Sisto SA, Dyson Hudson TA, Chang M, et al (2005). Pushrim biomechanics and injury prevention in spinal cord injury: recommendations based on CULP-SCI investigations. J Rehabil Res Dev 42: 9-19.
- Boninger, M. L., A. L. Souza, R. A. Cooper, S. G. Fitzgerald, A. M. Koontz and B. T. Fay (2002). Propulsion patterns and pushrim biomechanics in manual wheelchair propulsion. Archives of Physical Medicine and Rehabilitation 83(5): 718-723.
- Bregman DJJ, Drongelen Sv, Veeger HEJ (2009). Is effective force application in handrim wheelchair propulsion also efficient? Clinical Biomechanics 24: 13-19.
- Burnham RS, May L, Nelson E, Steadward R, Reid DC (1993). Shoulder pain in wheelchair athletes. The role of muscle imbalance. Am J Sports Med 21: 238-242.
- Chow JW, Levy CE (2011). Wheelchair propulsion biomechanics and wheelers' quality of life: an exploratory review. Disabil Rehabil Assist Technol 6: 365-377.
- Collinger JL, Boninger ML, Koontz AM, Price R, Sisto SA, et al (2008). Shoulder Biomechanics During the Push Phase of Wheelchair Propulsion: A Multisite Study of Persons With Paraplegia. Archives of Physical Medicine and Rehabilitation 89: 667-676.
- Curtis KA, Tyner TM, Zachary L, Lentell G, Brink D, et al (1999). Effect of a standard exercise protocol on shoulder pain in long-term wheelchair users. Spinal Cord 37: 421-429.
- Cowan RE, Nash MS, Collinger JL, Koontz AM, Boninger ML (2009). Impact of surface type, wheelchair weight, and axle position on wheelchair propulsion by novice older adults. Arch Phys Med Rehabil 90: 1076-1083.
- de Groot, S., H. E. Veeger, A. P. Hollander and L. H. van der Woude (2003). Adaptations in physiology and propulsion techniques during the initial phase of learning manual wheelchair propulsion. Am J Phys Med Rehabil 82(7): 504-510.
- de Groot, S., H. E. Veeger, A. P. Hollander and L. H. van der Woude (2004). Effect of wheelchair stroke pattern on mechanical efficiency. Am J Phys Med Rehabil 83(8): 640-649.
- Dyson-Hudson TA, Kirshblum SC (2004). Shoulder pain in chronic spinal cord injury, Part I: Epidemiology, etiology, and pathomechanics. J Spinal Cord Med 27: 4-17.
- Finley MA, Rasch EK, Keyser RE, Rodgers MM (2004). The biomechanics of wheelchair propulsion in individuals with and without upper-limb impairment. J Rehabil Res Dev 41: 385-395.
- Fullerton HD, Borckardt JJ, Alfano AP (2003). Shoulder pain: a comparison of wheelchair athletes and nonathletic wheelchair users. Med Sci Sports Exerc 35: 1958-1961.
- Gellman H, Sie I, Waters RL (1988). Late complications of the weight-bearing upper extremity in the paraplegic patient. Clin Orthop Relat Res: 132-135.
- Gutierrez DD, Thompson L, Kemp B, Mulroy SJ (2007). The relationship of shoulder pain intensity to quality of life, physical activity, and community participation in persons with paraplegia. J Spinal Cord Med 30: 251-255.
- Hosseini SM, Oyster ML, Kirby RL, Harrington AL, Boninger ML (2012). Manual wheelchair skills capacity predicts quality of life and community integration in persons with spinal cord injury. Arch Phys Med Rehabil 93: 2237-2243.
- Koontz AM, Cooper RA, Boninger ML, Souza AL, Fay BT (2002). Shoulder kinematics and kinetics during two speeds of wheelchair propulsion. J Rehabil Res Dev 39: 635-649.

Koontz, A. M., B. M. Roche, J. L. Collinger, R. A. Cooper and M. L. Boninger (2009).Manual Wheelchair Propulsion Patterns on Natural Surfaces During Start-UpPropulsion. Archives of Physical Medicine and Rehabilitation 90(11): 1916-1923.

- Kotajarvi BR, Basford JR, An KN (2002). Upper-extremity torque production in men with paraplegia who use wheelchairs. Arch Phys Med Rehabil 83: 441-446.
- Kwarciak AM, Turner JT, Guo L, Richter WM (2012). The effects of four different stroke patterns on manual wheelchair propulsion and upper limb muscle strain. Disability and Rehabilitation: Assistive Technology 7: 459-463.
- Lal S (1998). Premature degenerative shoulder changes in spinal cord injury patients. Spinal Cord 36: 186-189.
- LaPlante MP, Kaye HS (2010). Demographics and Trends in Wheeled Mobility Equipment Use and Accessibility in the Community. Assistive Technology 22: 3-17.
- Mercer JL, Boninger M, Koontz A, Ren D, Dyson-Hudson T, et al. (2006). Shoulder joint kinetics and pathology in manual wheelchair users. Clin Biomech (Bristol, Avon) 21: 781-789.
- Nichols PJ, Norman PA, Ennis JR (1979). Wheelchair user's shoulder? Shoulder pain in patients with spinal cord lesions. Scand J Rehabil Med 11: 29-32.
- Pentland WE, Twomey LT (1994). Upper limb function in persons with long term paraplegia and implications for independence: Part I. Paraplegia 32: 211-218.
- Rankin JW, Kwarciak AM, Richter WM, Neptune RR (2012). The influence of wheelchair propulsion technique on upper extremity muscle demand: a simulation study. Clin Biomech (Bristol, Avon) 27: 879-886.
- Rankin JW, Richter WM, Neptune RR (2011). Individual muscle contributions to push and recovery subtasks during wheelchair propulsion. J Biomech 44: 1246-1252.
- Richter, W. M., R. Rodriguez, K. R. Woods and P. W. Axelson (2007). Stroke Pattern and Hand-rim Biomechanics for Level and Uphill Wheelchair Propulsion at Self-Selected Speeds. Archives of Physical Medicine and Rehabilitation 88(1): 81-87.
- Sanderson, D. J. and H. J. Sommer, 3rd (1985). Kinematic features of wheelchair propulsion. J Biomech 18(6): 423-429.
- Shimada SD, Robertson RN, Bonninger ML, Cooper RA (1998). Kinematic characterization of wheelchair propulsion. J Rehabil Res Dev 35: 210-218.
- Silfverskiold J, Waters RL (1991). Shoulder pain and functional disability in spinal cord injury patients. Clin Orthop Relat Res: 141-145.
- Sinnott KA, Milburn P, McNaughton H (2000). Factors associated with thoracic spinal cord injury, lesion level and rotator cuff disorders. Spinal Cord 38: 748-753.
- van der Woude LH, de Groot S, Janssen TW (2006). Manual wheelchairs: Research and innovation in rehabilitation, sports, daily life and health. Med Eng Phys 28: 905-915.

Chapter 2

Experimental data collection procedures

This chapter describes the manual wheelchair propulsion data collection procedures and experimental configuration used in detail. The data analyses reported in Chapters 3, 4 and 5 are all based on this data.

Participants

Wheelchair propulsion data from 30 individuals between the age of 18 to 65 years, with a range of physical disabilities and a mix of genders (male (n=17) and female (n=13)) from the Urbana-Champaign campus community were collected. Inclusion criteria were: (1) between 18-65 years old and (2) use of a manual wheelchair as their primary means of mobility for 1+ year.

Participants were classified into "with shoulder pain" and "without shoulder pain" groups based on their self-report ("Yes"/"No"- written response) of shoulder pain to our demographic questionnaire provided at the time of data collection.

Experimental protocol

All experimental protocols in this study were approved by the University of Illinois at Urbana-Champaign institutional review board. Upon arrival to the laboratory, the experimental procedures were described to the participants and any questions they had regarding the protocol were clarified. Once participants understood the experimental procedures, they voluntarily signed the institutionally approved informed consent form. The participants then provided demographic information (age, height, weight, duration of wheelchair use, diagnosis, pain status, etc). In addition to self reporting their current status of shoulder pain ("Yes"/"No"), participants also rated their current level of shoulder pain on a 10 cm visual analog scale (VAS) (Campbell et al., 1990) and using the wheelchair user shoulder pain index (WUSPI) (Curtis et al., 1995).

A VAS score of 0 indicated that the participant was not experiencing any shoulder pain at the time of data collection and a score of 10 indicated existence of high level of shoulder pain at the time of data collection. The wheelchair user's shoulder pain index (WUSPI) is based on a 15-item questionnaire (Curtis et al., 1995). Each item is rated between 0 to 10, with 0 representing no interference with functional activities and 10 representing complete interference during the past week due to shoulder pain. The total score is the sum of scores of each of the 15 items. Total scores ranged from

0 (no pain) to 150 (maximum limitations to daily activities due to pain) (Curtis et al., 1995). Participants were separated into pain or no pain groups based on self-report of shoulder pain.

Following the voluntary disclosure of the self-reported shoulder pain scores the participant's, wheelchair configuration measurements and upper extremity anthropometry were measured. The wheelchair configuration measurements included, the shoulder X and Z coordinate positions with respect to the axle (XPOS and YPOS; Figure 2.1), camber, and wheel diameter. The upper extremity anthropometry measurements included torso length, upper arm length, forearm length, wrist circumference, elbow circumference, humerus circumference and knuckle circumference.

Following the collection of all volunteer demographic and shoulder pain data, the participants' personal wheelchair fitted bilaterally with 25 inch diameter force sensing SMARTWheels (Three Rivers Holdings LLC; AZ, USA). An individuals' upper extremity kinematics is not significantly affected by attaching to/testing with different SMARTWheel sizes (Mason et al., 2012). Attaching the SMARTWheels to the participant's personal wheelchair does not change the wheel placement alignment or camber (Mason et al., 2012). Each participant's wheelchair was then secured to a single drum dynamometer with a fly wheel and tie-down system (Figure 2.2). The use of force sensing wheels (Figure 2.2) allowed for the determination of temporal-

spatial and kinetic data relating to wheelchair propulsion. The reference axes orientations are shown in Figure 2.2.

Kinematic data collection: Motion Capture

Based on the International Society of Biomechanics (ISB) recommendations (Wu et al., 2005), 18 reflective markers were attached at specific bony landmarks to define the trunk, upper arm, forearm, hand, sternum and the jaw: these included sternal notch, C7 vertebrae, T3 vertebrae, T6 vertebrae and bilaterally at the mandible, third metacarpophalangeal joint, radial styloid ulnar styloid, olecronon, lateral epicondyl and the acromion process (Figure 2.2). Two reflective markers, one on the wheel center and the other on the wheel spoke were placed on each of the wheels (Figure 2.2). Kinematic data were collected using a 10 camera motion capture system (Cortex 2.5, Motion Analysis Co.; Santa Rosa, CA, USA) at a sampling rate of 100Hz.

Participants were asked to propel at constant speeds for three separate 3 minute trials at 1.1m/s (fast), 0.7m/s (slow) and self-select (~0.89 m/s) speeds. The sequence of trial speeds was randomized for each participant. A speedometer was used to provide real-time visual feedback to the subjects while kinetic data were collected bilaterally at 100Hz. Sufficient rest and recovery was provided between each trial. Subjects were

given time to acclimate to the dynamometer and propulsion speed before the beginning of each trial. A force plate was used to measure the weight of participants (AMTI, Inc., Watertown, MA, USA).

List of figures



Figure 2.1 The shoulder horizontal and vertical position with respect to the rear axle. XPOS: the horizontal distance between the rear axle and the shoulder, YPOS : the vertical distance between the rear axle and the shoulder.



Figure 2.2 Experimental setup in which data was collected for this dissertation. The motion capture marker placements, the instrumented SMARTWheel, the roller dynamometer setup with the flywheel and tie down system and the axes orientations are shown.

References

- Boninger ML, Koontz AM, Sisto SA, Dyson Hudson TA, Chang M, et al (2005). Pushrim biomechanics and injury prevention in spinal cord injury: recommendations based on CULP-SCI investigations. J Rehabil Res Dev 42: 9-19.
- Campbell WI, Lewis S (1990). Visual analogue measurement of pain. Ulster Med J 59: 149-154.
- Curtis KA, Roach KE, Applegate EB, et al (1995). Development of the Wheelchair User's Shoulder Pain Index (WUSPI). Paraplegia.;33:290–293.
- Mason BS, Van Der Woude LH, Tolfrey K, Lenton JP, Goosey-Tolfrey VL (2012). Effects of wheel and hand-rim size on submaximal propulsion in wheelchair athletes. Med Sci Sports Exerc 44: 126-134.
- Wu G, van der Helm FC, Veeger HE, Makhsous M, Van Roy P, et al (2005). ISB recommendation on definitions of joint coordinate systems of various joints for the reporting of human joint motion Part II: shoulder, elbow, wrist and hand. J Biomech 38: 981-992.

Chapter 3

Shoulder pain and jerk during wheelchair propulsion

Chapter abstract: Repetitive loading of the upper limb due to wheelchair propulsion plays a leading role in the development of shoulder pain in manual wheelchair users (mWCUs). There has been minimal inquiry on understanding wheelchair propulsion kinematics from a human movement ergonomics perspective. This investigation employs an ergonomic metric, jerk, to characterize the recovery phase kinematics of two recommended manual wheelchair propulsion patterns: semi-circular and the double loop. Further it examines if jerk is related to shoulder pain in manual wheelchair users. Data from 22 experienced adult mWCUs was analyzed for this study (semi-circular: n=12 (pain/without-pain:6/6); double-loop: n=10 (pain/without-pain:4/6)). Participants propelled their own wheelchair fitted with SMARTWheels on a roller dynamometer at 1.1 m/s for 3 minutes. Kinematic and kinetic data of the upper limbs were recorded. Three dimensional absolute jerk experienced at the shoulder, elbow and wrist joint during the recovery phase of wheelchair propulsion were computed. Twoway ANOVAs were conducted with the propulsion pattern type and shoulder pain as between group factors.

^{*}This chapter contains published work. The details are as follows: Chandrasekaran Jayaraman, Carolyn L. Beck, Jacob J. Sosnoff (2015). Shoulder pain and jerk during recovery phase of manual wheelchair propulsion. Journal of Biomechanics. In press. DOI: http://dx.doi.org/10.1016/j.jbiomech.2015.09.018.

Findings: (1) Individuals using a semi-circular pattern experienced lower jerk at their arm joints than those using a double loop pattern (*P*<0.05, $\eta^2=0.32$)_{wrist};(*P*=0.05, $\eta^2=0.19$)_{elbow};(*P*<0.05, $\eta^2=0.34$)_{shoulder} and (2) individuals with shoulder pain had lower peak jerk magnitude during the recovery phase (*P*≤0.05, $\eta^2=0.36$)_{wrist};(*P*≤0.05, $\eta^2=0.30$)_{elbow};(*P*≤0.05, $\eta^2=0.31$)_{shoulder}. *Conclusions:* Jerk during wheelchair propulsion was able to distinguish between pattern types (semi-circular and double loop) and the presence of shoulder pain. Jerk provides novel insights into wheelchair propulsion kinematics and in the future it may be beneficial to incorporate jerk based metrics into rehabilitation practice.

Introduction

Approximately 2 million Americans use a manual wheelchair for mobility (LaPlante et al., 2010). Although the use of a manual wheelchair provides numerous benefits (Hosseini et al., 2012), the repetitive strain encountered by the upper limb during propulsion places significant demand on the tissues (Nichols et al., 1979, Gellman et al., 1988, Curtiset al., 1999, Finley et al., 2004) and has been implicated in upper limb injury (Cooper et al., 1998). Indeed up to 70% of manual wheelchair users (mWCUs) report upper extremity pain (Finley et al., 2004). Upper extremity injury in mWCUs has been linked to difficulty performing activities of daily living, decreased physical activity and decreased quality of life (Chow et al., 2011).

Consequently, wheelchair propulsion research has led to guidelines to minimize over-use injuries (Cooper R et al 1998, Boninger et al., 2002, Koontz et al., 2002, Richter et al., 2007). In general the guidelines suggest that individuals use propulsion patterns such as semi-circular and double loop that maximize contact angle. However these guidelines do not discuss other kinematic markers of movement such as jerk, which has been implicated in overuse injuries (Berret et al., 2008, Srinivasan et al., 2012, William et al., 2008, Mark (2012)).

^{*}This chapter contains published work. The details are as follows: Chandrasekaran Jayaraman, Carolyn L. Beck, Jacob J. Sosnoff (2015). Shoulder pain and jerk during recovery phase of manual wheelchair propulsion. Journal of Biomechanics. In press. DOI: http://dx.doi.org/10.1016/j.jbiomech.2015.09.018.

Jerk, the third derivative of position has been widely employed in clinical rehabilitation and human motor control research to quantify movement smoothness and evaluate the performance of upper limb tasks (Hogan et al., 1987, Flash., 1990, Chang et al., 2005, Caimmi et al., 2008). Occupational ergonomics research has revealed distinct differences in arm jerk between movements in individuals with and without shoulder pain (Cote et al., 2005). Consequently, the purpose of this investigation is to examine jerk in wheelchair propulsion as a function of recovery pattern and shoulder pain.

To appreciate this research it is important that the reader understands that a typical push-rim wheelchair propulsion has two phases, a push phase (hands in contact with push-rim) and a recovery phase (hands move freely to initiate next push). Four general categories of recovery patterns widely reported in the literature are semi-circular (SC), single loop (SLOP), double loop (DLOP) and ARC (Shimada et al. 1998, Boninger et al., 2002, Richter et al., 2007, Raina et al., 2012, Slowik et al., 2015). The magnitude of forces and moments experienced by the shoulder joint during recovery phase of wheelchair propulsion can be as high as that during the push phase (Mercer et al., 2006, Sosnoff et al., 2015). Given this association, it is logical to expect that shoulder pain will influence arm kinematics during the recovery phase of wheelchair propulsion. Indeed recent research shows that mWCUs with shoulder pain employed spatial adaptive strategies to wrist kinematics

during the recovery phase of wheelchair propulsion (Jayaraman et al., 2014).

This analysis examines jerk-based metrics extracted from the three dimensional kinematics of the upper arm joints during wheelchair propulsion. The main goals are (1) to introduce and benchmark a jerk-based framework for wheelchair propulsion and (2) to examine jerk in wheelchair propulsion as a function of recovery pattern and shoulder pain. To accomplish these goals, two recovery pattern types, SC and DLOP patterns were analyzed. A DLOP pattern is characterized by the hands lifting over the propulsion path and crossing the propulsion path to drop below the hand-rim forming a double loop, while a SC pattern is characterized by the hands dropping below the hand-rim during the recovery phase (Sanderson et al., 1985, Shimada et al., 1988, Boninger et al., 2002, Richter et al., 2007).

We postulate two hypotheses: (H1) that individuals using a SC recovery pattern will experience lower jerk magnitudes at their wrists than individuals using a DLOP recovery pattern; (H2) that individuals with shoulder pain will minimize peak jerk magnitude at their upper arm joints during the recovery phase kinematics in an effort to avoid pain. H1 rests on the logical rationale that the arm's movement trajectory during a SC pattern is simpler than a DLOP pattern. H2 is based on the observation that the neuromuscular system avoids large acceleration changes to avoid pain (Berret et al., 2008).
Methods

Participant demographics

Wheelchair propulsion data from 22 experienced adult mWCUs were analyzed. This data constitutes a subset of data from a larger study (n=27) examining wheelchair propulsion and shoulder pain (Sosnoff et al., 2015) . The total number of participants that employed a SLOP (n=4) or ARC (n=1) were few, hence only SC and DLOP patterns were analyzed. Inclusion criteria for the larger investigation were: (1) between 18-65 years old and (2) use of a manual wheelchair as their primary means of mobility for 1+ year. Table 1 lists the participant demographic information. The recovery pattern types were classified using the third metacarpophalangeal joint's sagittal plane motion (Shimada et al., 1988). A sample DLOP and SC pattern are shown in Fig.1 (a1-b1) respectively. Twelve individuals used a SC pattern while ten participants used a DLOP pattern. These propulsion patterns were self-selected and no specific instructions regarding pattern were provided.

Kinematic and kinetic data collection

The kinematic and kinetic data collection procedures have been detailed in Chapter 2. Since participants switched the propulsion pattern used for different speeds, only the kinematics at 1.1 m/s speed was

analyzed.

Kinematic and kinetic data post-processing

The motion data were post- processed and any missing intermediate marker data points were fit using a cubic interpolation. This post-processing was accomplished with Cortex 2.5 Motion Analysis software. The hand-rim kinetic data from the SMARTWheel was used to identify the push and recovery phases from each propulsion cycle. The push phase's start and end points were located where the moment applied to the hand-rim (Mz) was greater and lower than 1 Nm, respectively, for at least 10 ms (Richter et al., 2011).

The post- processed motion data were filtered using a fourth-order low-pass Butterworth filter with 6 Hz cut-off frequency (Bednarczyk et al., 1994) . The wrist motion data was approximated as the mid-point of the radial styloid (RS) and ulnar styloid (US) hand segment marker coordinates. The elbow motion data was approximated as the mid-point of olecronon and lateral epicondyl, while the acromion process was used to represent the shoulder kinematics. To test our hypotheses (H1 & H2) we analyzed three dimensional motion data from wrist, elbow and shoulder joints.

Based on recommendation from previous literature, jerk metrics in this analysis were computed from the Cartesian coordinate motion data (Flash

(1990), Hreljac (2000)). From the three dimensional motion data (i.e. X, Y and Z coordinate) of the joints, the instantaneous resultant velocity along the trajectory was approximated. To accomplish this, first the individual velocity components from X,Y and Z coordinate motion data were calculated. Then the resultant of these individual velocity components was computed to obtain the instantaneous resultant velocity (Winter (2009)). The acceleration and jerk were approximated by obtaining the successive first and second order time derivatives of the resultant velocity respectively (Winter 2009). The absolute magnitude of the jerk was computed and used for further analyses. For all time derivative approximations, a two point central difference scheme was used (Winter 2009). The data was filtered using a fourth-order low-pass Butterworth filter with 6 Hz cut-off frequency before approximating each time derivative (Cooper et al., 2002, Winter (2009)).

To be consistent across individuals, 50 consecutively occurring recovery phases were extracted from each participant's data set. Each extracted absolute jerk curve was time normalized to 100 points using a shape preserving cubic spline. Two jerk measures were computed from these absolute jerk curves, namely a *jerk cost criteria* (J_c) and *peak jerk criteria* (PJ_c).

To compute J_c , first the area under the absolute jerk curve for each of the 50 extracted recovery cycles was computed. J_c was computed as the

average value of the area under the absolute jerk curve. The scheme used for computing J_c is shown in figure 3.3. The group-wise averaged J_c between the SC and DLOP groups were statistically compared to validate the first hypothesis. For this comparison, data belonging to the dominant hand side was analyzed (right side: n=19(SC=10; DLOP=9); left side: n=3(SC =2, DLOP=1)).

To compute PJ_c , first the P_{max} from each of cycle was extracted. P_{max} was defined to be the peak magnitude of absolute jerk that occurred during the recovery trajectory. Two distinct peak jerk magnitude locations were observed, the first between the 0% to 30% (see peak points P_1 in figure $3.2(c) \& P_4$ in figure 3.2(d) and the other between the 70% to 100% (see peak points P_2 in figure 3.2(c) & P_5 in figure 3.2(d)) intervals. $P_{J_c (0\% \text{ to } 30\%)}$ and $PJ_{c (70\% to 100\%)}$ were computed as the average of the peak magnitude of the absolute jerk (P_{max}) (averaged over the 50 consecutive cycles). The scheme used for computing PJ_c is shown in Figure 3.4. The group mean PJ_c 's from the two intervals were statistically compared between the groups with and without shoulder pain to validate the second hypothesis. The PJ_c belonging to the side of the hand with the greatest shoulder pain was analyzed for the pain group (right side: n=9 (SC=5; DLOP=4); left side: n=1(SC=1)). The PJ_c from the dominant hand side was used for the group without shoulder pain, (right side: n=10 (SC=5; DLOP=5); left side: n=2(SC=1; DLOP=1)).

Kinetic data processing

In addition to the kinematic data, the within individual cycle-wise spatial-temporal propulsion variables were extracted. These included, the mean contact angle, mean push time, mean push speed and mean peak resultant force at the hand-rim, each averaged over the 50 cycles considered. A custom developed MATLAB program was used for all computations.

Statistical analysis

All statistical data analyses were conducted using SPSS (version 21, IBM, Inc.). All values are reported as Mean (SD) unless otherwise noted. The significance level was set to $P \le 0.05$.

Independent variables

Participant demographics information (age, body weight, arm length and manual wheelchair propulsion experience) self-reported current level of shoulder pain (VAS scores) and WUSPI scores were treated as independent variables. A series of two tailed independent t-tests with propulsion pattern (SC or DLOP) and the shoulder pain (pain vs. no pain) as the between subject factors were conducted to check if statistically significant group differences existed in demographic variables. Two-tailed Mann–Whitney U

tests were used to identify if statistical significant difference in VAS and WUSPI scores existed between the pain groups.

Dependent variables

Mean contact angle, mean peak resultant force at hand-rim during push, mean push speed, Jc's and PJ_c's were treated as dependent variables. To test if statistically significant group differences existed in the dependent variables, a series of two-way analysis of variances with propulsion pattern (SC or DLOP) and the shoulder pain status (pain vs. no pain) as the between subject factors were conducted.

Results

Demographics

No statistically significant between group differences in demographics information as a function of recovery pattern type or shoulder pain status were observed, (P's>0.05; Table 3.1). Per design, the group with shoulder pain reported higher pain than the no pain group (VAS: [U=11,P<0.05]; WUSPI:[U=11,P<0.05])._No statistically significant difference in shoulder pain was observed as a function of recovery pattern (P>0.05).

Spatial-temporal propulsion variables at hand-rim

The group-wise mean (SD) of spatial-temporal propulsion variables are reported in Table 3.2. No statistically significant between group differences in peak resultant force, push speed or contact angle were observed (*P's*>0.05) as a function of pattern type. Push time was significantly different between the SC and DLOP groups $[F(1,18)=4.63,P<0.05,\eta^2=0.20]$ with the SC group having a greater push time. No statistical significant differences were observed in mean spatialtemporal propulsion variables as a function of shoulder pain (*P's*>0.05).

Recovery kinematic and jerk metrics

A representative plot of the resultant velocity, acceleration and jerk at the wrist for a DLOP and SC pattern are shown in figure 3.1 (a2-a4) and figure 3.1 (b2-b4), respectively. A time normalized (0% to 100%) absolute jerk curve computed for the wrist for SC and DLOP pattern are show in figure 3.2(c-d), respectively. Figure 3.3 (a-b) shows a sample area under the curve for SC and DLOP pattern types. The number of peak jerk points for a DLOP pattern appears higher than that for a SC pattern. The area under the curve for a DLOP pattern is larger than that of a SC pattern.

A sample recovery trajectory comparing peak jerk magnitudes (P_{max}) between the groups with and without shoulder pain for the SC and DLOP pattern types are shown in figure 3.4(a-b) respectively. It is clear from the sample data that irrespective of the pattern type, P_{max} magnitude for the individual with shoulder pain is lower than that for the individual without shoulder pain.

Jerk criteria (J_c)

A statistically significant main effect of recovery pattern (SC and DLOP) was observed for J_c at the wrist, elbow and shoulder joint; $[F(1,18)=8.49,P<0.05,\eta^2=0.32]_{wrist}$; $[F(1,18)=4.3,P=0.05, \eta^2=0.19]_{elbow}$; $[F(1,18)=9.28, P<0.05,\eta^2=0.34]_{shoulder}$. The SC group experienced lower mean J_c 's than the DLOP group during the recovery phase (See figure 3.5(ac)). No statistically significant between group differences in J_c was observed as a function of shoulder pain (P>0.05).

Peak jerk criteria (PJ_c)

A statistically significant main effect of shoulder pain was observed for $PJ_{c (0\% to 30\%)}$; $[F(1,18)=10.01, P<0.05, \eta^2=0.36]_{wrist}$; $[F(1,18)=7.8, P<0.05, \eta^2=0.30]_{elbow}$ and $[F(1,18)=8.16, P<0.05, \eta^2=0.31]_{shoulder}$. The shoulder pain group had lower $PJ_{c (0\% to 30\%)}$ magnitude at all the three joints

than the no pain group (See figure 3.6(a-c)). No statistically significant main effect of shoulder pain was observed for $PJ_{c(70\% to 100\%)}$ (*P*>0.05).

Discussion

In this investigation, the jerk characteristics of the upper limb during the recovery phase of manual wheelchair propulsion as a function of propulsion pattern and shoulder pain were examined. In agreement with our postulated hypotheses the SC recovery pattern experienced lower J_c and individuals with shoulder pain had less PJ_c regardless of propulsion style. Overall, our results suggest that, utilizing a jerk metric while analyzing manual wheelchair propulsion provides novel insights.

The mean spatial-temporal wheelchair propulsion parameters observed in this investigation were consistent with previous literature (Boninger et al., 1997, Shimada et al., 1998, Boninger et al., 2002, Collinger et al., 2008, Richter et al., 2011). This benchmarking was essential to suggest that the observations from our investigation are generalizable and qualitative comparison of our results with previous literature is acceptable.

The logical reason for the DLOP recovery pattern to incur greater J_c is a result of the joints undergoing sharp directional turns, leading to frequent

switching between acceleration and deceleration during the recovery trajectory. In contrast, when executing a SC pattern the arm undergoes less directional change resulting in lower jerk. Additionally, the relatively complex DLOP kinematics requires the upper extremity musculature to do additional work to overcome the inertia and gravity. From a jerk minimization perspective, it appears that the SC pattern appears to be superior.

The second novel observation from this investigation is that, individuals with shoulder pain minimized PJ_{c(0% to 30%)} at all the three joints, namely, the wrist, elbow and shoulder compared to the group without shoulder pain. This observation is consistent with research observation reported in occupational biomechanics. For instance, individuals with back pain lift a box with less jerk than those without back pain (Slaboda et al., 2005) and it was suggested that those with pain adopt a pain minimizing strategy, characterized by lower jerk. Another investigation revealed that shoulder pain influenced the kinematics of arm joints (Cote JN et al., 2005) with those with shoulder pain having a kinematic movement pattern with lower acceleration magnitudes than those without pain. In the context of our analysis, it is maintained that individuals with shoulder pain adopt a smoother arm motion pattern to reduce momentary discomfort at the shoulder during wheelchair propulsion.

Although the examination of jerk is relatively novel within wheelchair biomechanics research, it is consistent with human motor control research that has implemented jerk based measures to evaluate upper limb movement. Evidence from motor control research suggests that kinematic analysis involving jerk is a viable approach to quantify movement coordination during rehabilitation (Ramos et al., 1997, Teulings et al., 1997, Cozens et al., 2003, Caimmi et al., 2008, Chang et al., 2005). Our results and suggest that integrating jerk metrics with existing analysis procedures can yield an enhanced understanding of wheelchair propulsion mechanics.

For instance, wheelchair propulsion analyses have focused on studying the effect of the different recovery patterns on overall mechanical efficiency (Boninger et al., 2002, de Groot et al., 2004). There are divergent suggestions regarding the overall mechanical efficiency of SC and DLOP patterns (Shimada et al., 1998, Boninger et al., 2002, de Groot et al., 2004), with both being suggested as the more efficient propulsion technique. However, these studies did not differentiate the influence of different kinematic effects of the recovery phase jerk cost to their overall mechanical efficiency estimates. Perhaps integrating metrics such as jerk with existing wheelchair analysis procedures will yield novel information concerning mechanical efficiency.

Limitations

Despite being novel there are limitations that need to be acknowledged. Our sample size was small to investigate the influence of specific injury demographics on the jerk characteristics. However, the diversity of injury could also be viewed as a strength of this study. The results were significant despite having a sample with diverse injury demographics. The sample demographics limited our analysis to SC and DLOP patterns. It is not clear if similar movement characteristics could be identified in other recovery pattern types (ARC and SLOP; figure S1, S2 and S3). However for sake of completeness, the jerk characteristics for a sample SLOP and ARC patterns are provided in the chapter addendum section (Chapter addendum: Figures 3.7–3.9). Information on wrist pain was not collected. It is reasonable to expect that wrist pain is unlikely to influence kinematics during the recovery phase since the wrist experiences minimal forces/moments. A last limitation involves the laboratory based roller dynamometer setup utilized which provides a propulsion environment that does not exactly emulate real life propulsion.

Conclusions

This research implemented a novel approach integrating metrics and inferences from human movement ergonomics and motor control to understand kinematics of manual wheelchair users with shoulder pain. The analysis indicates that, adopting jerk based quantification of wheelchair

propulsion kinematics is worthwhile and yields insightful inferences. Overall the recovery phase kinematics of individuals using a SC recovery pattern placed lower jerk magnitudes than those using a DLOP and (2) mWCUs with shoulder pain had lower peak jerk magnitude during the recovery phase of wheelchair propulsion. In the future it may be beneficial to incorporate jerk based metric into rehabilitation practice.

List of figures



Figure 3.1 The instantaneous <u>resultant velocity</u>, acceleration and jerk at wrist from a sample DLOP (a1-a4) and SC (b1-b4) recovery pattern type. The solid lines in all the plots belong to the recovery phase and the dotted line to the push phase. The resultant velocity plot for wrist during the recovery for a DLOP (solid lines - a2) has two asymmetric velocity profile one for each loop as opposed to a SC pattern (solid line - b2).The rate of change of acceleration and deceleration for a DLOP (solid line - a3) is greater than a SC (solid line - b3) pattern. The jerk magnitude for the DLOP (solid line - a4) is greater than a SC (solid line - b4) pattern.



Figure 3.2 A sample time normalized (0% - 100% points) recovery trajectory absolute jerk curve at wrist for SC and DLOP patterns. (a) & (b) Push (dotted lines) and recovery (solid lines) phases for a SC and DLOP patterns respectively . The 0% and 100% points represent the start and end of the recovery phase. There are 100 data points between the 0% to 100% points; (c) & (d): Peak jerk magnitudes at wrist during SC and DLOP recovery patterns respectively. Peak jerk magnitude at wrist occurred between 0% to 30% (P1,P3,P4) and 70% to 100% (P2,P5) intervals along the recovery trajectory. These peak jerk points typically were seen to occur at those intervals along the recovery phase that required steep acceleration and deceleration rate of change.



where $\mathbf{C}_{\mathbf{m}}$ is the temporally located $\mbox{ mid-cycle of the trial.}$

Figure 3.3 The jerk cost criteria (J_c) : (a) & (b) are sample SC and DLOP pattern recovery phase absolute jerk curves depicting the area under the curve as shaded regions respectively. The scheme used for computing J_c is shown below the figures. C_m : the propulsion cycle located closest to the mid of the trial (i.e. 90 seconds from the start of trial).



where C_m is the temporally located mid-cycle of the trial.

Figure 3.4 The peak jerk criteria (PJ_c). (a) a sample plot from the SC group comparing $P_{max(0\%-30\%)}$ values between two individuals, with and without shoulder pain. (b) a sample plot from the DLOP group comparing $P_{max(0\%-30\%)}$ values between two individuals, with and without shoulder pain. The scheme used for computing P_{max} and PJ_c is shown below the figures. $C_{m is}$ the propulsion cycle located closest to the midpoint of the trial (i.e. 90 seconds from the start of trial).



Figure 3.5 Group mean comparison for J_c at the arm joints between SC and DLOP groups. (a) Group mean J_c at wrist joint for SC group is lower than that of the DLOP group; (b) Group mean J_c at elbow joint for SC group is lower than that of the DLOP group; (c) Group mean J_c at shoulder joint for SC group is lower than that of the DLOP group (*P \leq 0.05).



Figure 3.6 Group mean comparison for PJ_c at arm joints between groups with and without shoulder pain. (a) Group mean PJ_c at wrist joint for shoulder pain group is lower than that of the group without shoulder pain; (b) Group mean PJ_c at elbow joint for shoulder pain group is lower than that of the group without shoulder pain; (c) Group mean PJ_c at shoulder joint for shoulder pain group is lower than that of the group without shoulder pain group is lower than that of the group without shoulder pain; (c) Group mean PJ_c at shoulder joint for shoulder pain group is lower than that of the group without shoulder pain. (*P \leq 0.05)

List of tables

Table 3.1 Demographic information

	Recovery pattern type SC (n=12)		Shoulder pain sta With pain	tus No
Characteristics	Mean(SD)	DLOP (n=10) Mean(SD)	(n=10) Mean(SD)	pain(n=12) Mean(SD)
Age(years) Body weight (N) Arm length	25.5(10.68) 608.73(235.22)	21.5(4.1) 627.24(180.68)	25.8(11.11) 677.68(216.58)	22.00(5.31) 590.92(194.47)
(mm) Current shoulder pain	644.5 (54.51)	630.5 (33.76)	647(47.6)	630.9(45.8)
(VAS in cm)	2.11(3.22)	1.83(2.64)	4.16(3.00) *	0.02(0.09) *
WUSPI scores Experience using manual	14.58(22.77)	6.00(6.29)	20.70(22.28)*	2.33(3.75)*
wheelchair (Yrs)	16.83(5.72)	13.05(4.71)	16.45(6.18)	14.09(4.86)
Injury details	T6 and below (n=7); Sacral agenesis(n=1), Spinal cyst T6(n=1), Spina bifidia(n=2), C7(n=1).	T6 and below (n=5); Amputee(n=2), Spina bifidia (n=3).	T6 and below (n=5); Sacral agenesis(n=1), Spinal cyst T6(n=1), Spina bifidia(n=2), Amputee (n=1).	T6 and below (n=7); C7(n=1), Spina bifidia(n=3), Amputee (n=1).

Table 3.2 Mean (SD) of propulsion variables

	Recovery pattern type		Shoulder pain status	
Spatial-temporal propulsion variables	SC (n=12)	DLOP (n=10)	With pain (n=10)	No pain(n=12)
	Mean(SD)	Mean(SD)	Mean(SD)	Mean(SD)
Mean peak resultant force at hand-rim (N)	59.07 (18.38)	66.36 (20.81) 0.47	64.56(17.99)	60.57(21.11)
Mean push time (sec)	0.54 (0.06)*	(0.07)*	0.54(0.08)	0.48(0.06)
Mean contact angle (deg) Mean push speed	112.37 (12.21)	101.45 (13.30)	111.69(15.80)	103.84(11.26)
(m/sec)	1.12 (0.04)	1.17 (0.08)	1.12(0.05)	1.16(0.07)

*P<0.05

Chapter addendum



Figure 3.7 Wrist kinematic characteristics of a sample SLOP pattern: the hands rise above the hand rim during the recovery phase for a SLOP pattern; the dotted line lines in all the plots belong to the push phase and the solid lines to the recovery phase. Points P6,P7 and P8denote the jerk magnitude at the turns points made by the wrist during the SLOP recovery pattern.



Figure 3.8 Wrist kinematic characteristics of a sample ARC pattern: the hands follows the hand rim closely during the recovery phase for a ARC pattern forming a pumping action; the dotted line lines in all the plots belong to the push phase and the solid lines to the recovery phase. Points P 9and P 10 denote the jerk magnitude at the turns points made by the wrist during the ARC recovery pattern.



Figure 3.9 Group mean plots for $J_{\rm c}$ at the upper arm joints for SLOP and ARC patterns.

References

- Bednarczyk, J. H. and D. J. Sanderson (1994). Kinematics of wheelchair propulsion in adults and children with spinal cord injury. Arch Phys Med Rehabil 75(12): 1327-1334.
- Berret, B., C. Darlot, F. Jean, T. Pozzo, C. Papaxanthis and J. P. Gauthier (2008). The Inactivation Principle: Mathematical Solutions Minimizing the Absolute Work and Biological Implications for the Planning of Arm Movements. PLoS Comput Biol 4(10): e1000194.
- Boninger, M. L., R. A. Cooper, R. N. Robertson and T. E. Rudy (1997). Wrist biomechanics during two speeds of wheelchair propulsion: An analysis using a local coordinate system. Archives of Physical Medicine and Rehabilitation 78(4): 364-372.
- Boninger, M. L., A. L. Souza, R. A. Cooper, S. G. Fitzgerald, A. M. Koontz and B. T. Fay (2002). Propulsion patterns and pushrim biomechanics in manual wheelchair propulsion. Archives of Physical Medicine and Rehabilitation 83(5): 718-723.
- Boninger ML, Cooper RA, Fitzgerald SG, Lin J, Cooper R, Dicianno B, Liu B (2003). Investigating neck pain in wheelchair users.Am J Phys Med Rehabil.82(3):197-202.
- Caimmi, M., S. Carda, C. Giovanzana, E. S. Maini, A. M. Sabatini, N. Smania and F. Molteni (2008). Using Kinematic Analysis to Evaluate Constraint-Induced Movement Therapy in Chronic Stroke Patients. Neurorehabilitation and Neural Repair 22(1): 31-39.
- Campbell, W. I. and S. Lewis (1990). Visual analogue measurement of pain. Ulster Med J 59(2): 149-154.
- Chang, J.-J., T.-I. Wu, W.-L. Wu and F.-C. Su (2005). Kinematical measure for spastic reaching in children with cerebral palsy. Clinical Biomechanics 20(4): 381-388.
- Chow, J. W. and C. E. Levy (2011). Wheelchair propulsion biomechanics and wheelers' quality of life: an exploratory review. Disabil Rehabil Assist Technol 6(5): 365-377.
- Collinger, J. L., M. L. Boninger, A. M. Koontz, R. Price, S. A. Sisto, M. L. Tolerico and R. A. Cooper (2008). Shoulder biomechanics during the push phase of wheelchair propulsion: a multisite study of persons with paraplegia. Arch Phys Med Rehabil 89(4): 667-676.
- Cooper R, B. M. a. R. R. (1998). Heavy Handed: Repetitive strain injury among manual wheelchair users. Team Rehab Report 9(2): 35-38.
- Cooper RA, DiGiovine CP, Boninger ML, Shimada SD, Koontz AM and B. MA. (2002). Filter frequency selection for manual wheelchair biomechanics. J Rehabil Res Dev 39(3): 323-336.
- Cote JN, Raymond D, Mathieu PA, Feldman AG, Levin MF (2005). Differences in multi-joint kinematic patterns of repetitive hammering in healthy, fatigued and shoulder-injured individuals. Clin Biomech, 20 (6), pp. 581–590
- Cozens, J. A. and B. B. Bhakta (2003). Measuring movement irregularity in the upper motor neurone syndrome using normalised average rectified jerk. Journal of Electromyography and Kinesiology 13(1): 73-81.
- Curtis KA, Roach KE, Applegate EB, et al (1995). Development of the Wheelchair User's Shoulder Pain Index (WUSPI). Paraplegia.;33:290–293.
- Curtis, K. A., G. A. Drysdale, R. D. Lanza, M. Kolber, R. S. Vitolo and R. West (1999). Shoulder pain in wheelchair users with tetraplegia and paraplegia. Arch Phys Med Rehabil 80(4): 453-457.

- de Groot, S., H. E. Veeger, A. P. Hollander and L. H. van der Woude (2003). Adaptations in physiology and propulsion techniques during the initial phase of learning manual wheelchair propulsion. Am J Phys Med Rehabil 82(7): 504-510.
- de Groot, S., H. E. Veeger, A. P. Hollander and L. H. van der Woude (2004). Effect of wheelchair stroke pattern on mechanical efficiency. Am J Phys Med Rehabil 83(8): 640-649.
- de Groot, S., H. E. Veeger, A. P. Hollander and L. H. van der Woude (2004). Influence of task complexity on mechanical efficiency and propulsion technique during learning of hand rim wheelchair propulsion. Med Eng Phys 27: 41-49.
- Finley, M. A., E. K. Rasch, R. E. Keyser and M. M. Rodgers (2004). The biomechanics of wheelchair propulsion in individuals with and without upper-limb impairment. J Rehabil Res Dev 41(3b): 385-395.
- Flash, T. (1990). The Organization of Human Arm Trajectory Control. Multiple Muscle Systems. J. Winters and S.-Y. Woo, Springer New York: 282-301.
- Gellman, H., I. Sie and R. L. Waters (1988). Late complications of the weight-bearing upper extremity in the paraplegic patient. Clin Orthop Relat Res(233): 132-135.
- Hogan, N., E. Bizzi, F. A. Mussa-Ivaldi and T. Flash (1987). Controlling multijoint motor behavior. Exerc Sport Sci Rev 15: 153-190.
- Hosseini, S. M., M. L. Oyster, R. L. Kirby, A. L. Harrington and M. L. Boninger (2012).Manual wheelchair skills capacity predicts quality of life and community integration in persons with spinal cord injury. Arch Phys Med Rehabil 93(12): 2237-2243.
- Hreljac, A. (2000). Stride smoothness evaluation of runners and other athletes. Gait & Posture 11(3): 199-206.
- Jayaraman C, Moon Y, Rice IM, Hsiao Wecksler ET, Beck CL, et al. (2014). Shoulder Pain and Cycle to Cycle Kinematic Spatial Variability during Recovery Phase in Manual Wheelchair Users: A Pilot Investigation. PLoS ONE 9(3): e89794. doi:10.1371/journal.pone.0089794.
- Koontz, A. M., R. A. Cooper, M. L. Boninger, A. L. Souza and B. T. Fay (2002). Shoulder kinematics and kinetics during two speeds of wheelchair propulsion. J Rehabil Res Dev 39(6): 635-649.
- Koontz, A. M., B. M. Roche, J. L. Collinger, R. A. Cooper and M. L. Boninger (2009).Manual Wheelchair Propulsion Patterns on Natural Surfaces During Start-UpPropulsion. Archives of Physical Medicine and Rehabilitation 90(11): 1916-1923.
- LaPlante, M. P. and H. S. Kaye (2010). Demographics and Trends in Wheeled Mobility Equipment Use and Accessibility in the Community. Assistive Technology 22(1): 3-17.
- Mark RL, S. J. (2012). Introduction to human factors and ergonomics for engineers, CRC press.
- Mercer, Jennifer L. et al (2006). Shoulder joint kinetics and pathology in manual wheelchair users. Clinical Biomechanics , Volume 21 , Issue 8 , 781 789.
- Moon, Y., C. Jayaraman, I. M. Hsu, I. M. Rice, E. T. Hsiao-Wecksler and J. J. Sosnoff (2013). Variability of peak shoulder force during wheelchair propulsion in manual wheelchair users with and without shoulder pain. Clin Biomech (Bristol, Avon) 28(9-10): 967-972.
- Mulroy SJ, et al (2011) Strengthening and Optimal Movements for Painful Shoulders (STOMPS) in Chronic Spinal Cord Injury: A Randomized Controlled Trial. Phys Ther.:91:305-324.
- Nichols, P. J., P. A. Norman and J. R. Ennis (1979). Wheelchair user's shoulder? Shoulder pain in patients with spinal cord lesions. Scand J Rehabil Med 11(1): 29-32.

Ramos E, Latash MP, Hurvitz EA and B. SH. (1997). Quantification of upper extremity function using kinematic analysis. Arch Phys Med Rehabil 78(5): 491-196.

- Raina, S., McNitt-Gray, J., Mulroy, S., & Requejo, P. (2012). Effect of choice of recovery patterns on hand-rim kinetics in manual wheelchair users with paraplegia and tetraplegia. The Journal of Spinal Cord Medicine, 35(3), 148–155. http://doi.org/10.1179/2045772312Y.0000000013
- Raina, S., McNitt-Gray, J., Mulroy, S., & Requejo, P. (2012). Effect of choice of recovery patterns on hand-rim kinetics in manual wheelchair users with paraplegia and tetraplegia. The Journal of Spinal Cord Medicine, 35(3), 148–155. http://doi.org/10.1179/2045772312Y.0000000013
- Rankin JW, Richter WM, Neptune RR (2011). Individual Muscle Contributions to Push and Recovery Subtasks during Wheelchair Propulsion. Journal of biomechanics. 44(7):1246-1252.
- Rice IM, Jayaraman C, Hsiao-Wecksler ET and Sosnoff JJ (2013). Relationship Between Shoulder Pain and Kinetic and Temporal-Spatial Variability in Wheelchair Users. Archives of physical medicine and rehabilitation.
- Richter, W. M., A. M. Kwarciak, L. Guo and J. T. Turner (2011). Effects of single-variable biofeedback on wheelchair hand-rim biomechanics. Archives of Physical Medicine and Rehabilitation 92(4): 572-577.
- Richter, W. M., R. Rodriguez, K. R. Woods and P. W. Axelson (2007). Stroke Pattern and Hand-rim Biomechanics for Level and Uphill Wheelchair Propulsion at Self-Selected Speeds. Archives of Physical Medicine and Rehabilitation 88(1): 81-87.
- Sanderson, D. J. and H. J. Sommer, 3rd (1985). Kinematic features of wheelchair propulsion. J Biomech 18(6): 423-429.
- Shimada, S. D., R. N. Robertson, M. L. Bonninger and R. A. Cooper (1998). Kinematic characterization of wheelchair propulsion. J Rehabil Res Dev 35(2): 210-218.
- Slaboda, J. C., J. R. Boston, T. E. Rudy, S. J. Lieber and D. M. Rasetshwane (2005). The use of splines to calculate jerk for a lifting task involving chronic lower back pain patients. Neural Systems and Rehabilitation Engineering, IEEE Transactions on 13(3): 406-414.
- Slowik, Jonathan S. et al. (2015). The influence of speed and grade on wheelchair propulsion hand pattern, Clin Biomech (Bristol, Avon). 2015 Jul 21. pii: S0268-0033(15)00192-8. doi: 10.1016/j.clinbiomech.2015.07.007.
- Sosnoff JJ, Rice IM, Hsiao-Wecksler ET, Hsu IMK, Jayaraman C and Moon Y (2015). Variability in wheelchair propulsion: a new window into an old problem. Front. Bioeng. Biotechnol. 3:105. doi: 10.3389/fbioe.2015.00105.
- Srinivasan D, Mathiassen SE (2012). Motor variability in occupational health and performance. Clin Biomech (Bristol, Avon) 27: 979–993. doi: 10.1016/j.clinbiomech.2012.08.007.
- Teulings, H. L., J. L. Contreras-Vidal, G. E. Stelmach and C. H. Adler (1997).
 Parkinsonism reduces coordination of fingers, wrist, and arm in fine motor control.
 Experimental Neurology 146(1): 159-170.
- Veeger HEJ, van der Woude LHV, Rozendal RH (1991). Load on the upper extremity in manual wheelchair propulsion. Journal of Electromyography and Kinesiology 1: 270–280. doi: 10.1016/1050-6411(91)90014-v.
- William CW, R. F. (2008). Biomechanics of Musculoskeletal Injury, Human Kinetics.
- Winter, D. A. (2009). Biomechanics and Motor Control of Human Movement, Wiley.
- Wu, G., F. C. van der Helm, H. E. Veeger, M. Makhsous, P. Van Roy, C. Anglin, J. Nagels,A. R. Karduna, K. McQuade, X. Wang, F. W. Werner and B. Buchholz (2005). ISB

recommendation on definitions of joint coordinate systems of various joints for the reporting of human joint motion--Part II: shoulder, elbow, wrist and hand. J Biomech 38(5): 981-992.

Chapter 4

Shoulder pain and trunk kinematics in manual wheelchair propulsion

Chapter abstract: Trunk kinematics during wheelchair propulsion is known to influence propulsion biomechanics and also bears implication for shoulder injury. The main aim of this investigation was to study the trunk kinematic differences between manual wheelchair users with and without shoulder pain. Data from 25 experienced adult mWCUs was analyzed for this study (pain/without-pain:13/12)). Participants propelled their own wheelchairs fitted with SMARTWheels on a roller dynamometer at 1.1 m/s for 3 minutes. Kinematic and kinetic data of the upper limbs were recorded. The net drift in the trunk position (during steady state propulsion) from the initial reference position (rest) along the sagittal plane (X direction) was computed. Two-way ANOVA was conducted to investigate if statistically significant differences in trunk drift existed as a function of shoulder pain (p < 0.05). Overall our results show that the trunk drift along the sagittal plane for manual wheelchair users with shoulder pain was greater than those without pain $(p<0.05; n^2=0.30)$. Incorporating trunk kinematics based measures may provide additional knowledge to understand the adaptive strategies employed by mWCUs with shoulder pain. To our knowledge this is the first work documenting trunk kinematic differences between in mWCUs with and without shoulder pain.

Introduction

Approximately 70% of MWCUs report incidence of shoulder pain within the first 12 months of wheelchair use (Finley et al., 2004). Shoulder pain in MWCUs is reported to have significant negative impact on quality of life (Chow et al., 2011). Overall, any loss of upper limb function due to pain adversely impacts the independence and mobility of mWCUs. It has been speculated that a decrease in independence and mobility leads to greater health care costs and an increased risk for secondary morbidity (osteoporosis, obesity, cardiovascular disease etc) (Dyson-Hudson et al., 2004, Gellman et al., 1988., Geronda et al. 2004, Gutierrez et al., 2007, Finley et al., 2004). Consequently, over the past two decades many research investigations have focused on a plethora of biometric markers in an attempt to understand the relationship between shoulder injury and manual wheelchair propulsion. (Boninger et al., 2002, Collinger et al., 2008, Rankin et al., 2011, Rankin et al., 2012, Richter et al., 2007, 2011, Koontz et al., 20009, Koontz et al., 2012, de Groot et al., 2003, de Groot et al., 2004, Kwarciak et al., 2012, Mercer et al., 2006).

Evidence from this research highlight that the movement of the trunk during wheelchair propulsion is related to shoulder injury in MWCUs. Guidelines from these studies maintain that propelling a manual wheelchair with a 'trunk flexed forward/ anterior tilt' position exposes mWCUs to secondary injury risk (Rodgers et al., 2000, Rodgers et al., 2001, Rankin et

al., 2011, Gagnon et al., 2015, Gagnon et al., 2009, Sanderson DJ et al., 1985, Vanlandewijck YC et al., 1989, Chow et al., 2009, Yang et al., 2006, Rice et al., 2004).

Despite this large body of research, presently there is very limited information regarding trunk kinematics differences between groups propelling a manual wheelchair with and without shoulder pain. The is due to the reason that most of these studies on trunk kinematics in wheelchair propulsion literature studied user populations with healthy shoulder and/or a task constraint that may simulate a shoulder pain/exhaustion event (i.e. ramp, loaded propulsion). There is a need to extend this body of research to study the trunk kinematics of experienced mWCUs with shoulder pain. Our investigation seeks to directly address this void in the literature by studying the trunk kinematics differences between experienced manual wheelchair users with and without shoulder pain.

Consequently, this cross-sectional study analyzed the trunk kinematics during manual wheelchair propulsion in a group of experienced adult MWCUs with and without shoulder pain. We hypothesized that the trunk kinematics of individuals propelling a manual wheelchair with shoulder pain will significantly differ from the trunk kinematics of the group without shoulder pain. This being the first study to record trunk kinematic differences between users with and without shoulder pain, the outcome from this analysis will

potentially improve the knowledgebase and complement existing practices, leading to a better understanding of the role played by the trunk in the context of propulsion biomechanics and shoulder pain.

Methods

Protocol

Wheelchair propulsion kinetic and kinematic data from 25 individuals was analyzed in this study. Other aspects of this dataset have already been published (Sosnoff et al., 2015). Inclusion criteria for this secondary analysis were: (1) between 18-65 years old and (2) use of a manual wheelchair as their primary means of mobility for 1+ year and (3) Spinal related disability. Table 1 lists the participant demographic details. Reasons for wheelchair use included traumatic spinal cord injury (n=14), spinal cyst (n=1), spina bifidia(n=9), sacral agenisis (n=1).

Kinematic and kinetic data collection

The kinematic and kinetic data collection procedures have been detailed in Chapter 2. Since participants switched the propulsion pattern used for different speeds studying the kinematics at a specific speed was most appropriate. Consequently, only the data belonging to trials conducted with a speed of 1.1 m/s was used for this analysis. Before starting the trial, the participants were instructed to hold an upright sitting position (i.e. trunk held at 90 deg relative to the horizontal plane), for 5 seconds. The sternum motion marker position recorded for this upright sitting position was treated as the initial reference position for the trunk. Participants then pushed the wheelchair at a target speed of 1.1 m/s for 3 minutes. A speedometer was used to provide real-time visual feedback while kinetic data were collected bilaterally at 100Hz. Kinematic data were collected using a 10-camera motion capture system (Cortex 2.5, Motion Analysis Co.; Santa Rosa, CA, USA) at a sampling rate of 100Hz. Hand-rim force and moment data were collected by the SMARTWheel.

Kinematic data post-processing

The post-processing was carried out using Cortex 2.5 Motion Analysis software. During this phase any missing motion-marker data points were fitted using a cubic interpolation. Then the motion data was filtered using a 4th order low-pass Butterworth filter with a 6 Hz cut-off frequency (Morrow et al., 2011). The sternal notch motion data was used to characterize the trunk kinematics. The trunk kinematic metric derived from the sternum motion data in this analysis were computed with the wheel center as reference. Since our main variable of interest was the sagittal plane trunk excursion in X-direction (Figure 2.2), this analysis was based only on the X coordinate position data of the sternal notch.

The X coordinate position data of the sternal notch was extracted from each participant. Data beginning from the 6th propulsion cycle to the end of the trial was considered as steady state portion of a trial (Koontz et al., 2009, Jayaraman et al., 2014). The average value of the total sternal notch excursion (mm) in the X direction was computed. The absolute of the difference between the average value of the total sternal notch excursion (mm) and the initial reference position of the sternal notch was defined as T_{Drift} . Thus, T_{Drift} is a metric that captures the net drift in the trunk position over the entire trial period with respect to the initial reference position of the trunk, along the sagittal plane X direction.

The average trunk flexion angle during steady state propulsion was also computed. The computing procedure for 'average trunk flexion angle' was adopted from previous literature (Rodgers et al., 2001).

Kinetic spatial-temporal data extraction

The kinetic data from the SMARTWheel were used to identify the start and end of each propulsion cycle. The start and end points of each cycle were identified as points where the moment applied to the hand rim (Mz) was greater or lower than 1 Nm, respectively, for at least 10 ms (Jayaraman et al., 2014). All mean spatial-temporal wheelchair propulsion variables were extracted from the SMARTWheel hand-rim kinetic data.

A custom developed MATLAB code was used for the data post-

processing.

Statistical Analyses

All statistical data analyses were conducted using SPSS (version 21, IBM, Inc.). Descriptive statistics results are reported as mean(SD). For all statistical tests, the between group factor was shoulder pain status (i.e. pain vs. no pain). The significance levels for the statistical tests were set to $P \le 0.05$.

Independent variables

A series of two-tailed independent t-tests were conducted to verify if statistically significant differences existed in the independent variables between the groups. The independent variables included age, years of wheelchair propulsion experience, body weight, number of pushes and wheelchair axle position with respect to shoulder (XPOS and YPOS). Twotailed Mann–Whitney U tests were used to identify if statistically significant differences in VAS and WUSPI scores existed between the pain groups. The difference in gender composition between the pain/no pain groups was evaluated using a X^2 test.

Dependent variables

The mean values of spatial-temporal variables, namely, contact angle, push time, peak resultant push force at hand-rim, speed, trunk flexion and

 T_{Drift} were treated as dependent variables. A series of two-way ANOVA's were conducted to verify if statistically significant between-group differences existed in the mean spatiotemporal variables as a function of shoulder pain.

Results

Demographics

No significant between-group differences in independent variables, namely, age, years of wheelchair propulsion experience, body weight, and wheelchair axle positions, were observed as a function of shoulder pain (P's>0.05). The group with shoulder pain reported higher pain scores than the group without shoulder pain (VAS:[U=11, P<0.05]; WUSPI:[U=11, P<0.05]). Descriptive statistics of all the demographic variables are furnished in Table 4.1.

Mean spatial-temporal propulsion variables

No significant differences were observed in mean spatiotemporal propulsion variables as a function of shoulder pain (*P*'s>0.05). The groupwise mean (SD) of spatiotemporal propulsion variables are reported in Table 4.2. A significant main effect of shoulder pain was observed for T_{Drift} ; [F(1,23)=9.6, *P*<0.05, η^2 =0.30]. The group with shoulder pain had larger T_{Drift} (mm) than the group without shoulder pain (see Fig 4.3) [with shoulder pain: 46.5(25.1) mm; without shoulder pain: 22.3(10.5) mm].
Discussion

The relationship between shoulder pain and trunk kinematics during steady state manual wheelchair propulsion in a group of experienced adult MWCUs with and without shoulder pain was investigated. Overall in agreement with the postulated hypothesis the participants with shoulder pain had significantly larger T_{Drift} (mm) (i.e. larger trunk movement in the sagittal X direction) than the group without shoulder pain. This observation highlights that trunk kinematic metrics can differentiate between mWCUs with and without shoulder pain.

A logical explanation for the pain group to exhibit such differences in trunk kinematics could be that individuals with shoulder pain adopted a movement trunk strategy that minimized shoulder discomfort during the demanding propulsion task. This view is consistent with recommendations from occupation ergonomics literature (Madeleine et al., 2008, Lomond et al., 2010, Lomond et al., 2011). Individuals with neck/shoulder pain performing rhythmic repetitive occupational tasks adopted a spatial strategy manipulating the movement of their trunk to minimize discomfort (Madeleine et al., 2008, Lomond et al., 2010, Lomond et al., 2011).

The results holds significance because, in general a propulsion style with the trunk flexed forward has been reported to expose the arm joints of mWCUs to risk of injury (Rodgers et al., 2001, Raina et al., 2012). Moreover,

from a pure human movement ergonomics perspective, when the trunk is flexed forward, the arm joints grab the hand-rim with greater internal rotation at the initiation of the start of the push phase. Application of load to arm joints with such orientation is a well known to increase the risk of injury (Bridger RS 2009). Based on these observations, it is maintained that analyzing trunk kinematics in the context of shoulder pain in mWCUs is beneficial.

Previous wheelchair propulsion research included only a few propulsion cycles (~10-20) (Raina et al., 2012) to study trunk kinematics while our analysis uses the metrics computed over the entire trial (~135 cycles). The T_{Drift} metric in this work focuses on the net trunk excursion over the entire trial, rather than separating them as push/recovery portions. Another highlight that differentiates this work is that, the kinematic data was collected on individuals with shoulder pain. This is the first work documenting trunk kinematic differences between mWCUs with and without shoulder pain.

The mean values of the wheelchair propulsion variables obtained from our sample qualitatively compare with previous research (Collinger et al., 2008, Boninger et al., 2002). Similar to previous research there was no mean difference between group for the kinetic parameters (Collinger et al., 2008). The mean values for trunk flexion during push phase from our

sample matches with the values reported in wheelchair literature (Rodgers et al., 2001, Gagnon et al., 2015). Such benchmarking of the qualitative behavior of our data with previous research is essential to extend complementary inferences and findings.

Wheelchair configuration is known to influences the trunk kinematics during propulsion. No significant differences existed between groups for the wheelchair configuration variables (XPOS, YPOS). Based on this we maintain that the effect of wheelchair configuration differences between the pain groups were negligible.

Trunk control data for the different spinal injury level was not collected from the participants, which is a major limitation. However, the demographic spread in spinal injury levels between the groups with and without pain in our sample were very similar (Table 4.1). These exploratory results must be replicated with future research efforts using a sample with more homogenous injury demographics.

Limitations

Despite producing new results there are some limitations to this study. The study is cross sectional and cause/effect relationships cannot be inferred from these results. Trunk control data for the different spinal injury level was not collected from the participants. The pain scores were self-reported and

no radiographic/ultrasonic information regarding shoulder pain was collected. The sample size was small. Data was collected in a laboratory environment and it is not clear if the results are replicable in real life environment.

Conclusions

Trunk kinematics differences between individuals with and without shoulder pain was studied. The net drift in trunk X coordinate position from the initial references position was calculated over the entire trial duration. Our observation highlights that trunk kinematic metrics can differentiate between mWCUs with and without shoulder pain. Individuals with shoulder pain had larger net drift in trunk position than those without shoulder pain (p<0.05). Incorporating trunk kinematics based measures, may provide additional knowledge to understand the adaptive strategies employed by mWCUs with shoulder pain. To our knowledge this is the first work documenting trunk kinematic differences between mWCUs with and without shoulder pain.

List of figures



Figure 4.1 A sample trunk Sagittal plane motion (X kinematics) data. The initial reference position, the start-up push phases and the steady state portions are shown in the magnified view.



Figure 4.2 Sample trunk (X coordinate) kinematics for participants with and without shoulder pain. The data is for the X coordinate motion of the trunk for a 3 minute trial. It can be obsrerved that for th individual with pain, the trunk position drift is larger than the individual without pain.



Figure 4.3 Group mean T $_{\rm Drift}$ between groups with and without shoulder pain. The group with shoulder pain had higher net T $_{\rm Drift}$

List of tables

Table 4.1 Demographic variables

Demographic variables	No Pain	Pain
	Mean(SD)	Mean(SD)
Age (Yrs)†	20.1(2.0)	28.3(12.5)
Body weight(Kg)	55.5(15.68)	75.63(24.71)
Wheelchair experience (yrs)	13.7(4.3)	15.6(11.0)
XPOS (mm)	3.1(53.8)	17.4(27.6)
YPOS (mm)	660.1(60.8)	695.8(57.4)
VAS [†]	0.1(0.4)*	4.5(2.5)*
WUSPI†	2.7(4.9)*	23.9(20.3)*
Camber (deg)	3(1.5)	2.0(1.9)
Gender (F / M) Injury levels	6 / 6 SB (n=5); T12 complete (n=1); T6 (n=2); T9 incomplete (n=1); Spinal AVM T6-T9 (n=1); L4 (n=1); Arthrogryposis (n=1)	6 / 7 SB (n=4); T12 complete (n=1); T6 (n=2); T8 incomplete (n=2); Birth defect T11 to L2 (n=1); L3 incomplete (n=1);L1 (n=1); Sacral Agenisis (n=1)

† Non parametric test

* P<0.05

Table 4.2 Mean wheelchair propulsion variables

Mean propulsion variables	No Pain	Pain Moon(SD)
	Mean(SD)	mean(SD)
Contact angle (deg)	98.4(19.7)	98.2(19.5)
Peak resultant force at hand-		
rim	55.1(16.01)	69.3(23.28)
Mean push speed (m/s)	1.13(0.05)	1.11(0.04)
Trunk flexion angle (deg)	3.3(0.6)	3.3(0.8)
Number of pushes	139.2(29.3)	145(31.6)
* P<0.05		

References

- Boninger ML, Koontz AM, Sisto SA, Dyson Hudson TA, Chang M, et al (2005) Pushrim biomechanics and injury prevention in spinal cord injury: recommendations based on CULP-SCI investigations. J Rehabil Res Dev 42: 9-19.
- Chow JW, Levy CE (2011) Wheelchair propulsion biomechanics and wheelers' quality of life: an exploratory review. Disabil Rehabil Assist Technol 6: 365-377.

Collinger JL, Boninger ML, Koontz AM, Price R, Sisto SA, et al. (2008) Shoulder biomechanics during the push phase of wheelchair propulsion: a multisite study of persons with paraplegia. Arch Phys Med Rehabil 89: 667-676.

- Dyson-Hudson TA, Kirshblum SC (2004) Shoulder pain in chronic spinal cord injury, Part I: Epidemiology, etiology, and pathomechanics. J Spinal Cord Med 27: 4-17.
- Finley MA, Rasch EK, Keyser RE, Rodgers MM (2004) The biomechanics of wheelchair propulsion in individuals with and without upper-limb impairment. J Rehabil Res Dev 41: 385-395.
- Dany Gagnon, Mary C.V, Kei M, Sylvie N, Rachid A and Milos R.P (2009). Effects of trunk impairments on on manual wheelchair propulsion among individuals with SCI: A brief overview and future challenges. Top Spinal Cord Inj Rehabil; 15(2):59-70.
- Dany Gagnon, Annie-Clude Badineau, Audrey Champagne et al (2015). Trunk and shoulder kinematics and kinetics and EMG adaptations to slope increase during motorized treadmill propulsion among manual wheelchair users with spinal cord injury. BioMed Research International. Article ID : 636319.
- Gellman H, Sie I, Waters RL (1988) Late complications of the weight-bearing upper extremity in the paraplegic patient. Clin Orthop Relat Res: 132-135.
- Gironda RJ, Clark ME, Neugaard B, Nelson A (2004) Upper limb pain in a national sample of veterans with paraplegia. J Spinal Cord Med 27: 120-127.
- Gutierrez DD, Thompson L, Kemp B, Mulroy SJ (2007) The relationship of shoulder pain intensity to quality of life, physical activity, and community participation in persons with paraplegia. J Spinal Cord Med 30: 251-255.
- Jayaraman C, Moon Y, Rice IM, Hsiao Wecksler ET, Beck CL, et al. (2014) Shoulder Pain and Cycle to Cycle Kinematic Spatial Variability during Recovery Phase in Manual Wheelchair Users: A Pilot Investigation. PLoS ONE 9(3): e89794. doi:10.1371/journal.pone.0089794.

Koontz AM, Cooper RA, Boninger ML, Souza AL, Fay BT (2002) Shoulder kinematics and kinetics during two speeds of wheelchair propulsion. J Rehabil Res Dev 39: 635-649.

- Koontz, A. M., B. M. Roche, J. L. Collinger, R. A. Cooper and M. L. Boninger (2009).
 Manual Wheelchair Propulsion Patterns on Natural Surfaces During Start-Up
 Propulsion. Archives of Physical Medicine and Rehabilitation 90(11): 1916-1923.
- LaPlante MP, Kaye HS (2010) Demographics and Trends in Wheeled Mobility Equipment Use and Accessibility in the Community. Assistive Technology 22: 3-17.
- Lomond KV, Cote JN (2010) Movement timing and reach to reach variability during a repetitive reaching task in persons with chronic neck/shoulder pain and healthy subjects. Exp Brain Res 206: 271-282.
- Lomond KV, Cote JN (2011) Differences in posture-movement changes induced by repetitive arm motion in healthy and shoulder-injured individuals. Clin Biomech (Bristol, Avon) 26: 123-129.

- Madeleine P, Mathiassen SE, Arendt Nielsen L (2008) Changes in the degree of motor variability associated with experimental and chronic neck-shoulder pain during a standardised repetitive arm movement. Exp Brain Res 185: 689-698.
- Madeleine P, Voigt M, Mathiassen SE (2008) The size of cycle-to-cycle variability in biomechanical exposure among butchers performing a standardised cutting task. Ergonomics 51: 1078-1095.
- Mercer JL, Boninger M, Koontz A, Ren D, Dyson-Hudson T, et al. (2006) Shoulder joint kinetics and pathology in manual wheelchair users. Clin Biomech (Bristol, Avon) 21: 781-789.
- Melissa M.B Morrow, Kenton R Kaufman and Kai-Nan (2011). Scapula kinematics and associated impingement risk in manual wheelchair users during propulsion and a weight relief lift. Clin Biomech; 26(4):352-357.
- Raina, S., McNitt-Gray, J., Mulroy, S., & Requejo, P. (2012). Effect of increased load on scapular kinematics during manual wheelchair propulsion in individuals with paraplegia and tetraplegia. Human movement science, 31, 397–407. http://doi.org/10.1179/2045772312Y.0000000013.
- Rankin JW, Richter WM, Neptune RR. Individual Muscle Contributions to Push and Recovery Subtasks during Wheelchair Propulsion. Journal of biomechanics. 2011;44(7):1246-1252.
- Mary M Rodgers, Randall E Keyser, Elizabeth K Rasch, Peter H Gorman et al (2001). Influence of training on biomechanics of wheelchair propulsion. JRRD volume 38(5); 505-511.
- Mary M Rodgers, Randall E Keyser, Elizabeth K Rasch, Peter H Gorman et al (2001). Influence of trunk flexion on biomechanics of wheelchair propulsion. JRRD, Vol 37(3); 283-295.
- Vanlandewijck YC, Spaepen AJ, Lysens RJ (1994) Wheelchair propulsion efficiency: movement pattern adaptations to speed changes. Med Sci Sports Exerc 26: 1373-1381.
- Yang YS, Koontz AM, Triolo RJ, Mercer JL, Boninger ML (2006). Surface electromyography activity of trunk muscles during wheelchair propulsion. Clin Biomech (Bristol, Avon); 21(10):1032-1041.

Chapter 5

Relationship between cycle-to-cycle structure in variability and shoulder pain in manual wheelchair users

Chapter abstract: The use of a manual wheelchair places considerable repetitive mechanical strain on the upper limbs leading to shoulder pain. While recent research indicates that the amount of variability in wheelchair propulsion variables is related to shoulder pain, there has been minimal enquiry in studying the relation between shoulder pain and the time dependent structure of variability in wheelchair propulsion variables. Data from 27 experienced adult manual wheelchair users with and without shoulder pain from varying disability demographics was analyzed. Participants propelled their own wheelchair fitted with SMARTWheels on a roller dynamometer at 1.1 m/s for 3 minutes. Sample entropy of cycle to cycle fluctuations in contact angle and the inter push time interval was compared between the groups with and without shoulder pain using nonparametric test. The main findings were: (1) variability observed in the fluctuation in contact angle during manual wheelchair propulsion is structured (Z=3.15,p<0.05); (2) individuals with shoulder pain exhibited higher sample entropy for contact angle during wheelchair propulsion compared to those without pain ($\chi^2(1) = 6.12$, p = 0.013); and (3) the SampEn measure correlated significantly with the amount of self-reported shoulder pain ($r_{s(WUSPI)} = 0.41$; $r_{s(VAS)} = 0.56$; p<0.05). Overall results show that

studying the time dependent structure in variability provides novel knowledge for tracking and monitoring shoulder pain in manual wheelchair users.

Introduction

Motor variability provides unique information concerning the control and health of the neurophysiologic systems (Lipsitz et al., 2004, Sosnoff et al., 2006). Researchers have shown that both the amount and the time dependent dynamical structure of motor variability provides useful insights about the control and function of the neurophysiologic systems (Newell, 1993, Sosnoff et al., 2006). The amount and structure of motor variability are distinct in that they can be uniquely influenced by experimental or population factors (Sosnoff et al., 2006). Moreover, dynamical measures of motor variability have been found to be more sensitive to pathology than distributional statistics (Slifkin et al., 1999, Stergiou et al., 2011). It is maintained that a lack of variation results in insufficient time to adapt (i.e. heal) between loading occasions.

The importance of examining the structure of variability can be seen with a brief review of locomotion research. A detailed examination of locomotion reveals that there are subtle structured (i.e. non-random) variations in the timing between each step (i.e. inter-stride interval) (Hausdorff et al., 1997). The structure of stride intervals in ambulatory walking have been shown to be influenced by pathology, developmental age, speed, and coordination pattern (walking vs. running) (Jordan et al., 2008). Currently, it is maintained that the dynamic structure of walking and running result from the complex interaction of supraspinal inputs to the spinal motor

neurons (Newell, 1993, Sosnoff et al., 2006, Lipsitz et al., 2004).

A unique form of locomotion is manual wheelchair propulsion. Although, an estimated 2 million individuals in the United States propel manual wheelchairs as their primary form of locomotion (LaPlante et al., 2010) very little motor control research has focused on it. The majority of investigations of wheelchair propulsion have examined mean performance variables and not explicitly focused on the variability characteristics of propulsion (Boninger et al., 2002, Collinger et al., 2008, Rankin et al., 2011, Richter et al., 2007, 2011, Koontz et al., 2009, 2012, de Groot et al., 2003,2004, Kwarciak et al., 2012).

Recent research indicates that variability in wheelchair propulsion mechanics is related to shoulder pain (Sosnoff et al., 2015). This association between motor variability and pain is consistent with observations from occupation biomechanics and human motor control literature (Srinivasan et al.,2012, Madeleine et al., 2009, Hamill et al., 2012, Stergiou et al., 2011).

Although promising, a limitation of research by Sosnoff et al., 2015, is that it has only focused on the amount of variability in wheelchair propulsion variables. This approach seemingly ignores the time-dependent structure inherent in motor output (Lipsitz et al., 1992).

Based on the tenets of the loss of complexity hypothesis of aging (Lipsitz et al., 2004) it has been theorized that the time-dependent structure of motor output is a marker of physiological complexity and provides novel information concerning the health of the musculoskeletal system (Sosnoff et al., 2006, Stregiou et al., 2011). Specifically, it has been proposed that musculoskeletal injury leads to motor fluctuations that are more structured. For instance, Tochigi et al., 2012 utilized sample entropy to analyze the structure of variability in gait in individuals with knee osteoarthritis with and without pain. Consistent with the loss of complexity hypothesis, it was reported that the pain group had significantly lower SampEn values when compared to those without pain. In a similar fashion, researchers have successfully employed other time dependent measures such as approximate entropy to study the gait pattern related to anterior cruciate ligament (ACL) injury (Georgoluis et al., 2006) and reported lower approximate entropy values in the knee with ACL deficiency.

Presently there is no information regarding the time dependent structure of variability in the context of shoulder pain in mWCUs. Consistent with the tenets of the *loss of complexity* hypothesis it is hypothesized that individuals with shoulder pain will demonstrate lower complexity compared to those without shoulder pain. Consequently the purpose of this investigation is to examine if the variable structure in wheelchair propulsion parameters is related to shoulder pain.

Methods

Participants

Wheelchair propulsion data from 27 experienced adult mWCUs was analyzed for this study. This dataset is a subset of data collected for a larger study focusing on wheelchair propulsion and shoulder pain (Sosnoff et al., 2015). Inclusion criteria for this secondary analysis were (1) more than one year of manual wheelchair experience; (2) between 18-64 years of age; and (3) trials with a minimum 106 wheelchair propulsion data cycles. All procedures were approved by the local institutional review board at University of Illinois, Urbana-Champaign.

Kinematic and kinetic data collection

The kinematic and kinetic data collection procedures have been detailed in Chapter 2. Since participants switched the propulsion pattern used for different speeds, studying the kinematics at one specific speed (i.e.1.1 m/s) was most appropriate. Consequently, only the data belonging to trials conducted with a speed of 1.1 m/s was used for this analysis.

Data post-processing

SMARTWheel data were collected at a sampling rate of 100Hz and digitally filtered with an eighth-order, zero-phase, low-pass Butterworth filter with 20Hz cutoff frequency (Collinger et al., 2008). The start and end of a

propulsion cycle was defined when the push-rim moment (Mz) was above and below 1 Nm respectively (Jayaraman et al., 2014). To reduce the transient effects, data belonging to the first five propulsion cycles were not included for this analysis (Jayaraman et al., 2014). For consistency, the number of data cycles analyzed for each participant was maintained constant at 100 cycles (i.e. starting from the 6th cycle to 105th cycle of a SMARTWheel data, figure 5.1(a)). The contact angle and resultant forces at the hand-rim were extracted for each participant (Figure 5.1(b)). Following this, the intrapush time interval between peak resultant force were extracted (Figure 5.1(b)). Thus, this process yielded two time series from each participant wheelchair propulsion dataset namely, (1) a time series of cycle-to-cycle contact angle (Figure 5.2(a)) and (2) a time series of the cycle-to-cycle intra-push time interval between peak resultant force during push phase (Figure 5.2(b)). Sample entropy was then computed for these two time series (see below). A custom developed MATLAB code was used to accomplish all data post-processing. Based on the VAS scores, for the shoulder pain group, the data belonging to the side with greatest shoulder pain level was analyzed (right (n=11) and left (n=2)), while the data belonging to the dominant hand (right (n=12) and left (n=2)) was analyzed for the group without shoulder pain.

Structure in variability

Complexity analysis quantifies the time dependent regularity and predictability of a time series. There are various nonlinear dynamics tools to measure the complexity of a physiological time series (Stergiou et al., 2004). The choice of a specific tool depends on the characteristics of data being analyzed. In this investigation, we utilized sample entropy (SampEn), a widely utilized approach (Yentes et al., 2013, Tochigi et al., 2012).

SampEn is a nonlinear metric used for measuring the complexity of a time series. The magnitude of the SampEn is an indication of the regularity or irregularity of a particular time series. SampEn values typically ranges from 0 to 4. Values closer to 0 are consistent with greater regularity, such as a sine wave, while values nearing 4 represent greater irregularity such as pink noise (Richman et al., 2000). Higher values of SampEn indicate that the time series is more unpredictable (i.e. unstructured).

Details of the sample entropy algorithms and input parameters used are available elsewhere (Richman et al., 2004; Lake et al., 2002). A MATLAB codes obtained from <u>http://www.physionet.org/physiotools/</u> was used for computing SampEn (Goldberger et al., 2000). The performance (SampEn magnitudes) of the software program was tested with synthetically generated signals (Chapter addendum material: Section A) to ascertain the validity of the code. The tests revealed that the SampEn values obtained

from using the software program provided by physionet.com is comparable with those reported in literature (Chapter addendum: Section A).

The parameters chosen for the SampEn analysis were (m=2, r=0.2) for the contact angle and (m=2,r=0.15) for the inter-push interval. The justifications for these parameter choices are provided in the Chapter addendum: Section B.

From each participant's post-processed data, the SampEn for the original time series belonging to the contact angle and the within individual inter-push time interval between peak resultant force was calculated. To statistically establish if the SampEn measure obtained from the original time series was indeed structured and not random, a simple surrogate analysis was carried out (Shelhamer, 2006).

Surrogate data sets were generated by randomly shuffling the original time series in order to determine whether the original time series was structured (Figure 5.3). Such random shuffling preserves the distributional statistics (i.e., mean, standard deviation, and higher moments) between the original time series and its corresponding surrogate time series pool as they contain the same elements; however, the sequential ordering is destroyed (Jordan et al., 2008). Each participant's original time series was randomly shuffled 100 times to produce a pool of corresponding surrogate time series for

each original time series. The sample entropy values were computed for each of the 100 surrogate time series in the pool and then averaged to generate a mean sample entropy for the surrogate pool (Figure 5.3). Finally non-parametric pair-wise tests (p<0.05) comparing the SampEn values between each original time-series with the corresponding mean sample entropy obtained from its surrogate pool was conducted. If the sample entropy value of the original time series was significantly (statistically) smaller than its corresponding surrogate counterpart, then this indicated that the original data was not randomly derived and therefore its variability may be structured (Hausdroff et al., 1997, Newell et al., 1993, Dingwell et al., 2000).

The SampEn magnitudes can be influenced by non-stationarity of the time series (Yentes et al., 2013, Slifkin et al., 1999). To check for this influence, the SampEn for both the spatial and temporal variables were calculated for the differenced time series. No significant differences were found between the SampEn calculated from the original and the differenced times series (Chapter addendum: Section C). Thus the SampEn from the original time series were used for statistical analyses.

Statistical Analysis

IBM SPSS (version 21) was used for conducting all the statistical tests. The significance level for all the statistical tests were set at $p \le 0.05$. All

descriptive values are reported as mean (SD) unless otherwise noted. The between group factor for all the statistical tests was set as shoulder pain (with pain = 1; No pain = 0).

Normality checks for data distribution using Shapiro-Wilk tests revealed that the age, self-reported WUSPI scores, mean inter push time interval, mean speed and the SampEn required non-parametric statistical tests.

Between group comparisons for independent variables

Participant demographics information (age, body weight, and manual wheelchair propulsion experience in years) and self-reported scores (WUSPI & VAS) were treated as independent variables.

A series of two tailed Mann-Whitney U tests were conducted to check if demographic variables (age and self-reported shoulder pain (WUSPI & VAS)) were significantly different between groups. A series of two tailed independent t-tests were conducted to check if significant between group differences in existed in body weight and wheelchair propulsion experience. Separate Chi-squared tests were conducted to check if category variables (gender and propulsion pattern type) were significantly different between the groups with and without shoulder pain.

Mean wheelchair propulsion variables

A series of two tailed Mann-Whitney U tests were conducted to check if the mean inter push time interval and mean push speed were significantly different between groups with and without shoulder pain. A series of two tailed t-tests were performed to check if statistically significant group differences existed in other mean wheelchair propulsion variables (i.e. contact angle, peak resultant force at hand-rim).

Pair-wise comparisons between SampEn original and surrogate data

To test if the SampEn of the original time series were significant different from their surrogate counterpart, a series of pair-wise Wilcoxon signed rank test were conducted for the SampEn of contact angle and inter push time.

Shoulder pain and SampEn of spatiotemporal variables

To investigate the main effect of shoulder pain on SampEn, the mean SampEn obtained from the original time series for the contact angle and inter-push time interval were compared between the groups using Kruskal-Wallis ANOVA's.

Spearman's rank correlational analyses were conducted to investigate if the SampEn for the contact angle and time interval was correlated with the self-reported pain scores (WUSPI; VAS).

Results

Demographics

No significant group differences in demographics were observed, (*P*'s>0.05; Table 1). Per design the shoulder pain group reported higher pain than the no shoulder pain group (WUSPI: [U=17,P<0.05]; VAS: [U=14.5,p<0.05]; Table 5.1).

Mean wheelchair propulsion variables

No significant differences were observed in mean spatial-temporal propulsion variables between groups (P's > 0.05; Table 5.2).

SampEn comparison between original and surrogated time series

Wilcoxon signed rank pair wise tests revealed that the SampEn obtained from the original time series were statistically smaller than the SampEn from their corresponding surrogates : inter push time interval : $[Z=2.59,p=0.009]_{(m=2,r=0.15)}$; contact angle: $[Z=3.15,p=0.002]_{(m=2,r=0.2)}$. This indicates that the variability structure found in contact angle and inter push interval variable in wheelchair propulsion are not random but rather structured.

Shoulder pain and structure of propulsion variability

The SampEn for contact angle was higher in the pain group than the no-pain group (Figure 5.4(a)). This was confirmed by Kruskal-Wallis test, $\chi^2(1) = 6.12$, p = 0.013. No significant main effect of shoulder pain group was observed for the SampEn of inter-push time interval (Figure 5.4(b); p>0.05).

A statistically significant moderate positive correlation was observed between the WUSPI and VAS scores and the SampEn of contact angle [rs(25)=0.41, p<0.05]WUSPI; [rs(25)= 0.56, p<0.05]VAS. There was no association between SampEn of inter push time interval and self reported pain scores (p>0.05).

Discussion

This investigation explored the relation between shoulder pain and the structure of motor variability in spatiotemporal variables of manual wheelchair propulsion. Three novel observations were made, (1) variability observed in the fluctuations in contract angle and inter push time interval during manual wheelchair propulsion is structured, (2) individuals with shoulder pain exhibited higher SampEn during wheelchair propulsion compared to those without pain; and (3) SampEn measures correlated significantly with the amount of self reported shoulder pain. Overall the observations suggest that SampEn analysis of wheelchair propulsion may

provide novel insights for monitoring the development and treatment of shoulder pain in mWCUs.

While the majority of wheelchair research focuses on mean kinematic and kinetics during propulsion (Boninger et al., 2002, Collinger et al., 2008, Rankin et al., 2011, Richter et al., 2007, 2011, Koontz et al., 20009, 2012, de Groot et al., 2003,2004, Kwarciak et al., 2012), growing evidence suggests that variability measures could be a sensitive marker of shoulder pain in mWCUs (Sosnoff et al., 2015). To date, the few wheelchair propulsion studies which incorporate measures of intra-individual variability have exclusively focused on linear distributional statistics (standard deviation and coefficient of variation) and not measures of time dependent structure.

The current investigation used SampEn , to quantify the structure in two propulsion parameters closely implicated with shoulder pain in mWCUs, namely, contact angle and inter-push time interval of peak resultant force, (Boninger et al., 2002, Collinger et al., 2008, Rankin et al., 2011, Richter et al., 2007, 2011, Koontz et al., 20009, 2012, de Groot et al., 2003, Kwarciak et al., 2012). Surrogate analyses revealed that the time dependent variable structure observed in both measures was not random. This observation is consistent with reports from occupational ergonomics and motor control research that time dependent structure in variability of physiologic output may offer meaningful insights concerning health and function (Madeleine et al., 2008, Madeleine et al., 2009, Vaillancourt et al.,2002, Vieluf et al., 2015).

It is important to note that differences in wheelchair users with and without shoulder pain in SampEn were only observed in fluctuations in contact angle. This is distinct from examinations of the magnitude of variability which revealed that individuals with shoulder pain had greater amounts of variability in spatial and temporal variability as indexed by coefficient of variation (Rice et al., 2014). The divergent results between time-dependent and magnitude metrics are consistent with the view that they are at least partially independent (Sosnoff et al., 2004, Sosnoff et al., 2006).

The observation that SampEn was higher for the shoulder pain group is in agreement with occupation ergonomics research (Madeleine et al., 2009). For instance, individuals with shoulder/neck pain performing repetitive upper limb tasks exhibited greater complexity in movement than healthy controls (Madeleine et al., 2008, Madeleine et al., 2009, Lomond et al., 2010, Lomond et al., 2011). Such time dependent structure in the movement has been implicated as compensatory strategies adopted by the neuromuscular system to mitigate the discomfort arising from musculoskeletal pain (Madeleine et al., 2009, Srinivasan et al., 2012). Along these lines, the higher SampEn magnitude observed for the contact angle at the hand-rim in mWCUs with shoulder pain could be a manifestation of their compensatory strategies to minimize shoulder discomfort when performing the repetitive propulsion task.

Consistent with previous research where higher discomfort (pain) levels were associated with higher spatial complexity (Madeleine et al., 2009), SampEn of contact angle was positively correlated with the selfreported pain scores. This observation suggests that, in addition to being able to differentiate between mWCUs with and without shoulder pain, SampEn may also be a potential biomarker to track shoulder pain progression in mWCUs. This exciting possibility warrants further investigation.

It is important to note that the current observations are counter to the original hypothesis. The original hypothesis that individuals with shoulder pain would have lower complexity in their wheelchair propulsion was based on the tenets of the *loss of complexity* hypothesis. The loss of complexity hypothesis maintains that with senescence and pathology there is a decline in physiological complexity resulting from reductions in control process and their interaction (Goldberger et al., 1992). A criticism of this theoretical framework is that it does not take into the intrinsic dynamics of the motor task (Vaillancourt et al., 2002; Sosnoff et al., 2006). Indeed previous research demonstrates that observed differences in complexity between pathological groups can be bi-directional and their directionality is influenced task constraints studied (Vaillancourt et al., 2002, Sosnoff et al., 2002, Sosnoff et al., 2006, Vieluf et al., 2015).

Finally, while SampEn is one among the many available non-linear dynamics metrics, numerous researchers have used other techniques like de-trended fluctuation analysis (DFA) and Lyapunov exponent (Ly Ep) to study the time dependent structure in human gait (Cavanaugh et al., 2006,Hausdroff et al., 1997, Stergiou et al., 2011, Sethi et al., 2013). All these studies reported that physiologic time series complexity is sensitive to age, pathology and functional state in humans. It is important to note that the current observations concerning SampEn do not necessarily indicate existence of any long range correlations. Nevertheless, outcome from our study is a first step towards showing that pursuing further research in this direction could be beneficial and improve our knowledge to provide better health/diagnosis/intervention to prevent shoulder pain in mWCUs.

Limitations

Despite being novel this study is not devoid of limitations. The study is cross-sectional in nature and hence any cause/effect inference cannot be draw. The choice of metrics utilized was hampered by the short time series length which prevented other complementing non-linear analyses such as de-trended fluctuation analysis. In general, adopting a combination of nonlinear methods is recommended as best practice (Stergiou 2004). The data were collected in a laboratory setting with roller dynamometer setup, so it is not clear if these results would occur in real life propulsion environment. It could be possible that complexity measures of spatiotemporal propulsion

variables other than ones considered in this analyses are more sensitive to shoulder pain in mWCUs. This requires further analysis. The injury demographics of our sample was diverse. Consequently, it is possible that differences in complexity between pain groups were due to different disability demographics. Though significant, these demographic and environmental limitations are relatively common to wheelchair propulsion research. Finally, despite these limitations, this investigation provides novel contributions that are important and compliments the research findings from previous research (Sosnoff et al., 2015, Madeleine et al., 2009, Srinivasan et al., 2012).

Conclusions

This cross sectional study provided novel information regarding the relationship between the time-dependent structure of wheelchair propulsion and shoulder pain in mWCUs. Examining the structure of wheelchair propulsion using sample entropy as an index of complexity, this investigation produced three novel observations, namely, (1) variability observed in the fluctuation in contract angle during manual wheelchair propulsion is structured, (2) individuals with shoulder pain exhibited greater complexity as indexed by SampEn during wheelchair propulsion compared to those without pain; and (3) SampEn measure correlated significantly with the amount of self-reported shoulder pain. The higher sample entropy in shoulder pain group may be a compensatory mechanism adopted to minimize discomfort

to shoulder while performing the propulsion task. Overall, we conclude that incorporating non-linear dynamics based measures in wheelchair propulsion analyses may provide new knowledge and can be an important tool for better health, diagnosis, and therapeutic interventions to prevent shoulder pain pathology in mWCUs.

List of figures



Figure 5.1. Data processing and variable extraction for SampEn calculation. (a) resultant force time series at hand-rim showing the steady state portion of the time series used for all calculations; (b) a magnified view of two sample resultant force cycles to show the details of the variables extracted. The contact angle is defined as the angular measure between the start and end of contact. The time interval between peak to peak resultant forces was extracted from cycle-to-cycle during steady state to extract the time series for inter push time interval. The cycle-to-cycle contact angle magnitude during the steady state propulsion was extracted to create the time series capturing fluctuations in contact angle.



Figure 5.2. Sample contact angle and inter push time interval time series extracted for SampEn calculation. (a) the cycle-to-cycle contact angle magnitude at hand-rim for 100 pushes; (b) the cycle-to-cycle inter push time interval between peak resultant force at hand-rim for 100 pushes.

Surrogates of original time series (i=100)

Original time series



Figure 5.3. The surrogate analysis process. The mean(SD) of the original and the surrogated time series pool are same. The surrogation procedure used was a simple random shuffling of data. A pool of 100 randomly shuffled surrogates was created from the original time series. If $(SampEn)_{original}$ is statistically lesser (p<0.05) than $(MeanSampEn)_{Surrogate}$, then the time dependent structure observed in the original time series is not random and may have some meaningful insight.



Figure 5.4 Groupwise box plot comparison for SampEn. (a) the SampEn for contact angle was significantly greater (p<0.05) for the group with shoulder pain; (b) the SampEn for inter push time interval at hand-rim between groups with and without shoulder pain was not significantly different (p>0.05) between the groups with shoulder pain.

List of tables

Table 5.1 Demographic information

Demographic variable	Pain (n=13)	No Pain (n=14)
	Mean(SD)	Mean(SD)
Age (Yrs)	28.23(12.52)	21.21(4.92)
Body weight (LB) Experience using wheelchair	164.50(56.39)	131.17(36.96)
(Yrs)	15.84(11.25)	13.64(5.15)
Gender (F/M)	6/7	6/8
		Spina Bifida (n=4)
	Spina Bifida(n=5)	T6-T12 (n=6)
	T6-T12 (n=5)	L1-L4 (n=1)
	L1-L4 (n=1)	Amputee-single
	Amputee- double	(n=1)
	(n=1)	Arthrogryposis
Injury demographics	Sacral agenisis (n=1)	(n=1) C7 (n=1)
WUSPI*	22.84(21.27)	3.28(5.02)
Table 5.2 Mean propulsion variables at hand-rim

Mean propulsion parameters at hand-rim	Pain (n=13) Mean(SD)	No Pain (n=14) Mean(SD)
Peak resultant force (N) Contact angle (deg) Mean speed (m/s)	69.82(23.84) 100.52(20.25) 1.1(0.04)	60.23(19.72) 97.68(17.64) 1.1(0.06)
time interval between peak resultant force (Sec)	1.15(0.22)	1.20(0.22)

Chapter addendum: Selection of sample entropy parameters

A. Checking the validity of the SampEn algorithm from physionet.com

To check the validity of the SampEn code from physionet.com, its output was benchmarked with Sample entropy values reported for standard theoretical time series in the literature (Richmann et al., 2000, Lake et al., 2002, Yentes et al., 2013). Five time series with 100 points were created, namely, (1) a periodic sinusoid; (2) a sinusoid with additive white noise; (3) a random white noise signal; (4) a periodic logistic map, and (5) chaotic logistic map. All time series were generated using custom code written in MATLAB. The details of the schemes used for generating these theoretical time series are shown in figure 5.5. The SampEn values obtained using the physionet.com code for a range of r values for each of these theoretical signals are show in figure 5.6. The SampEn magnitudes for the theoretical signals are consistent with those reported in literature (Richmann et al., 2000, Lake et al., 2002, Yentes et al., 2013). Similar to previously reported literature, the SampEn values dropped for increasing r values for a fixed m=2 (Yentes et al., 2013). Based on these observations, the SampEn code from physinet.com was deemed suitable to be used for our analyses.

Signal type	Scheme used for generation	Parameters	Generated signals
Periodic Sinusoid	Y=A*Sin(αt)	A = Amplitude; α=circular frequency (2*π*f); t=time sample length.	$\begin{array}{c} 1 \\ 0.5 \\ 0 \\ -0.5 \\ -1 \\ 0 \\ 20 \\ 40 \\ 60 \\ 80 \\ 100 \end{array}$
Sinusoid+ white_noise	S=Y+ (N _A * white_noise)	Y=A*Sin(αt); (N _A * white_noise) = additive white noise. Where, N _A = White noise amplitude; white_noise term generated using (randn) command in MATLAB. (randn-normally distributed pseudo random white noise)	
Random white noise	S=rand	uniformly distributed random white noise was generated using (rand) command in MATLAB.	$= \begin{bmatrix} 2 \\ 1 \\ 0 \\ -1 \\ -2 \\ -3 \\ 0 \end{bmatrix} = \begin{bmatrix} 2 \\ 0 \\ 20 \end{bmatrix} = \begin{bmatrix} 2 \\ 40 \end{bmatrix} = \begin{bmatrix} 2 \\ 60 \end{bmatrix} = \begin{bmatrix} 2 \\ 80 \end{bmatrix} = \begin{bmatrix}$
Periodic logistic map	x _{n+1} =a*(x _n)*(1-x _n)	a=3.4 and initial condition x _n =0.5. xn is a real number between (0-1)	$\begin{array}{c} 0.9 \\ 0.8 \\ 0.7 \\ 0.6 \\ 0.5 \\ 0.5 \\ 0.4 \\ 0 \\ 20 \\ 40 \\ 60 \\ 80 \\ 100 \\$
Chaotic logistic map	x _{n+1} =a*(x _n)*(1-x _n)	a=4 and initial condition x_n =0.99. xn is a real number between (0-1)	1 0.8 0.6 0.4 0.2 0 20 40 60 80 100

Figure 5.5 Schemes used to generate the synthetic signals for SampEn calculations. The X axes in all plots denote the number of time sample points (100 points). The number of time series points was chosen to match with the number of points analyzed in the data. Y axes denote the amplitude for each signal.



Figure 5.6 SampEn values obtained for the theoretical time series using the code from physionet.com. For all the calculations a m value of 2 and r values ranging from 0.1 to 0.3 at steps of 0.05 were used. (a) shows the SampEn for a regular sinusoid and a sinusoid with additive white noise for a range of r values; (b) SampEn for a random white noise signal; (c) shows the SampEn for a regular logistic map and a chaotic logistic map for a range of r values. Values and trends are consistent with previous literature (Yentes et al., 2013).

B. Sample entropy (SampEn) parameter selection

B1. Justification for selection of the embedding dimension (m)

For all SampEn calculations the embedding dimension (m) used was m=2. We offer two rationales for choosing m=2, namely, (1) since our focus was to capture the temporal evolution of cycle to cycle variability, a value of m=2 was chosen (m=2 signifies that the dynamics between the adjacent pairs of events are studied during the template similarity matching process) (Yentes et al., 2013) and (2) from a numerical perspective for reliable SampEn results, the recommended thumb rule relation between the time series length and embedding dimension (m) is usually $10^m - 20^m$ (Richmann et al., 2000, Lake et al., 2002). Since the length of the time series we studied had 100 points, based on the above recommendations, a value of m=2 (i.e. 100 time series points= 10^2) and range of r values, [r=0.15, 0.2, 0.25 and 0.3] where m is the embedding dimension and r is the tolerance (Richman et al., 2004; Lake et al., 2002, Yentes et al., 2013) were most appropriate for our dataset.

The ranges for r values for m=2, were decided based on previous research on human movement, which recommended r values greater than 0.1 when m=2 (Yentes et al., 2013). The appropriate r was chosen based on the relative consistency of the SampEn values observed in the dataset (Yentes et al., 2013).

B2. Justification for selecting the tolerance (r) values

Based on recommendations from literature, a range of r values were tested before picking the final r value. The ranges chosen were [r=0.1,0.15, 0.2, 0.25 and 0.3]. These ranges were picked based on recommendations for human movement data (Yentes et al., 2013). Based on the SampEn trends from the range of r value, the appropriate tolerances (r) for the spatial and temporal variables were picked.

In order to pick the correct r value, the relative consistency of the SampEn magnitudes between the pain and no pain groups were checked. A particular r is termed as 'relative inconsistent' if there occurs a switch in SampEn magnitude direction between the pain/no pain groups when compared to the trends obtained from the for the preceding/proceeding r value (Pincus et al., 1994, Yentes et al., 2013). It was observed that the relative magnitude of the group mean SampEn for the inter push time interval switched direction between the pain/no pain groups at a parameter choice of r=0.25 (Figure 5.7). It can be observed from figure.B1, that at r=0.2, the pain group demonstrated higher SampEn magnitude than the group without pain. At r=0.25 this trend switched direction with the no pain group having higher SampEn magnitude (Figure 5.7). Due to this relative inconsistency, the r values around 0.25 were deemed as not ideal choices (i.e. r values 0.2,0.25 and 0.3 are not ideal) (Yentes et al., 2013). Based on this observation an r value of 0.15 was chosen for the inter push time

interval).

The SampEn trends for contact angle variable was very consistent between group during all the ranges of r value studied (Figure 5.8). Following this, an r=0.2 was chosen for the spatial variable in-order to be consistent with human movement literature (Yentes et al., 2013) thus enabling benchmarkability. Consistent with previously reported literature, the SampEn magnitudes for both, the contact angle and the inter push time interval decreased with increasing r values (Yentes et al., 2013).



Figure 5.7 Selecting r value for the inter push time interval. The SampEn values between the pain/ no pain groups for different r values. At r=0.25 a relative inconsistency in trends observed when compared to the SampEn magnitude trends at r=0.2.



Figure 5.8 Selecting r value for the contact angle. The SampEn values between the pain/ no pain groups for different r values. At r=0.25 a relative inconsistency in trends observed when compared to the SampEn magnitude trends at r=0.2.

C. <u>Check for Non-stationary time series effect to SampEn</u>

Non-stationary time series data can lead to spurious SampEn value, leading to wrong inferences. To avoid this, the SampEn between the original and a differenced version of the original time series were computed for both the spatial and temporal variables. Wilcoxon signed rank pair wise tests revealed that the there were no statistically significant difference in SampEn values obtained from the original and the differenced time series (p>0.05; group mean(SD) shown in Table 5.3).

Table 5.3 Group	mean (SD)	values of S	SampEn	between	the o	original	and	the
differenced time	series							

Variables for which SampEn were computed	Between group factor	Group mean (SD) of SampEn for original time series	Group mean (SD) of SampEn for differenced time series	m	r
Spatial (Contact	No Pain	1.8(0.23)	1.8(0.27)	2	0.2
angle)	Pain	2.1(0.24)	2.0(0.13)	2	0.2
Temporal	No Pain	2.2(0.41)	2.2(0.40)	2	0.15
(Time interval)	Pain	2.5(0.60)	2.5(0.60)	2	0.15

References

- Boninger ML, Koontz AM, Sisto SA, Dyson Hudson TA, Chang M, et al (2005) Pushrim biomechanics and injury prevention in spinal cord injury: recommendations based on CULP-SCI investigations. J Rehabil Res Dev 42: 9-19.
- Bregman DJJ, Drongelen Sv, Veeger HEJ (2009) Is effective force application in handrim wheelchair propulsion also efficient? Clinical Biomechanics 24: 13-19.
- Campbell, W. I. and S. Lewis (1990). Visual analogue measurement of pain. Ulster Med J 59(2): 149-154.
- Cavanaugh, J. T., Guskiewicz, K. M., Giuliani, C., Marshall, S., Mercer, V. S., & Stergiou,
 N. (2006). Recovery of postural control after cerebral concussion: New insights using approximate entropy. Journal of Athletic Training, 41(3), 305-313.
- Collinger JL, Boninger ML, Koontz AM, Price R, Sisto SA, et al (2008) Shoulder Biomechanics During the Push Phase of Wheelchair Propulsion: A Multisite Study of Persons With Paraplegia. Archives of Physical Medicine and Rehabilitation 89: 667-676.
- Collinger, J. L., M. L. Boninger, A. M. Koontz, R. Price, S. A. Sisto, M. L. Tolerico and R. A. Cooper (2008). Shoulder biomechanics during the push phase of wheelchair propulsion: a multisite study of persons with paraplegia. Arch Phys Med Rehabil 89(4): 667-676.
- Curtis KA, Roach KE, Applegate EB, et al (1995) Development of the Wheelchair User's Shoulder Pain Index (WUSPI). Paraplegia.;33:290–293.
- Curtis, K. A., G. A. Drysdale, R. D. Lanza, M. Kolber, R. S. Vitolo and R. West (1999). Shoulder pain in wheelchair users with tetraplegia and paraplegia. Arch Phys Med Rehabil 80(4): 453-457.
- Curtis, K. A., G. A. Drysdale, R. D. Lanza, M. Kolber, R. S. Vitolo and R. West (1999). Shoulder pain in wheelchair users with tetraplegia and paraplegia. Arch Phys Med Rehabil 80(4): 453-457.
- Dalyan M, Cardenas DD, Gerard B (1999) Upper extremity pain after spinal cord injury. Spinal Cord 37: 191-195.
- De Groot S, Veeger DH, Hollander AP, Van der Woude LH (2002) Wheelchair propulsion technique and mechanical efficiency after 3 wk of practice. Med Sci Sports Exerc 34: 756-766.
- de Groot S, Veeger HE, Hollander AP, van der Woude LH (2004) Effect of wheelchair stroke pattern on mechanical efficiency. Am J Phys Med Rehabil 83: 640-649.
- DINGWELL, J.B. and CUSUMANO, J.P., 2000. Nonlinear time series analysis of normal and pathological human walking. Chaos, 10(4), pp. 848.
- A.D. Georgoulis, C. Moraiti, S. Ristanis, N. Stergiou (2006). A novel approach to measure variability in the anterior cruciate ligament deficient knee during walking: the use of the approximate entropy in orthopaedics. J Clin Monit Comput, 20 (1) (2006), pp. 11–18
- Gironda RJ, Clark ME, Neugaard B, Nelson A (2004) Upper limb pain in a national sample of veterans with paraplegia. J Spinal Cord Med 27: 120-127.
- Goldberger AL, Amaral LAN, Glass L, Hausdorff JM, Ivanov PC, Mark RG, Mietus JE, Moody GB, Peng CK, Stanley HE:PhysioBank, PhysioToolkit, andPhysioNet: Components of a New Research Resource for complex Physiologic Signals.Circulation2000,101(23):e215- e220 [http://circ.ahajournals.org/cgi/content/full/101/23/e215]

- Gutierrez DD, Thompson L, Kemp B, Mulroy SJ (2007) The relationship of shoulder pain intensity to quality of life, physical activity, and community participation in persons with paraplegia. J Spinal Cord Med 30: 251-255.
- Hamill J, Palmer C, Van Emmerik RE. Coordinative variability and overuse injury. Sports Med Arthrosc Rehabil Ther Technol. 2012 Nov 27;4(1):45.
- Hamill J, van Emmerik RE, Heiderscheit BC, Li L (1999) A dynamical systems approach to lower extremity running injuries. Clin Biomech (Bristol, Avon) 14: 297-308.
- Hausdorff, J. M., Mitchell, S. L., Firtion, R., Peng, C. K., Cudkowicz, M. E., Wei, J. Y., & Goldberger, A. L. (1997). Altered fractal dynamics of gait: Reduced stride-interval correlations with aging and huntington's disease. Journal of Applied Physiology, 82(1), 262-269.
- Jayaraman C, Moon Y, Rice IM, Hsiao Wecksler ET, Beck CL, et al. (2014) Shoulder Pain and Cycle to Cycle Kinematic Spatial Variability during Recovery Phase in Manual Wheelchair Users: A Pilot Investigation. PLoS ONE 9(3): e89794. doi:10.1371/journal.pone.0089794.
- Jordan, K. And Newell, K.M., 2008. The structure of variability in human walking and running is speed-dependent. Exercise and sport sciences reviews, 36(4), pp. 200-204.
- Koontz AM, Cooper RA, Boninger ML, Souza AL, Fay BT (2002) Shoulder kinematics and kinetics during two speeds of wheelchair propulsion. J Rehabil Res Dev 39: 635-649.
- Koontz, A.M., Worobey, L.A., Rice, I.M., Collinger, J.L. and Boninger, M.L., 2012.Comparison Between Overground and Dynamometer Manual Wheelchair Propulsion.Journal of Applied Biomechanics, 28(4), pp. 412-419.
- Kwarciak AM, Turner JT, Guo L, Richter WM (2012) The effects of four different stroke patterns on manual wheelchair propulsion and upper limb muscle strain. Disability and Rehabilitation: Assistive Technology 7: 459-463.
- Lake, D.E., Richman, J.S., Griffin, M.P., Moorman, J.R., 2002. Sample entropy analysis of neonatal heart rate variability. American Journal of Physiology: Regulatory, Integrative & Comparative Physiology, 52(3), pp. R789.
- Lal S (1998) Premature degenerative shoulder changes in spinal cord injury patients. Spinal Cord 36: 186-189.
- Lamoth CJ, Daffertshofer A, Meijer OG, Beek PJ (2006) How do persons with chronic low back pain speed up and slow down? Trunk-pelvis coordination and lumbar erector spinae activity during gait. Gait Posture 23: 230-239.
- LaPlante, M. P. and H. S. Kaye (2010). Demographics and Trends in Wheeled Mobility Equipment Use and Accessibility in the Community. Assistive Technology 22(1): 3-17.
- Lipsitz LA (2004) Physiological complexity, aging, and the path to frailty. Sci Aging Knowledge Environ 2004: pe16.
- Lipsitz LA, Goldberger AL (1992). Loss of 'complexity' and aging. Potential applications of fractals and chaos theory to senescence.
- Lomond KV, Cote JN (2010) Movement timing and reach to reach variability during a repetitive reaching task in persons with chronic neck/shoulder pain and healthy subjects. Exp Brain Res 206: 271-282.
- Lomond KV, Cote JN (2011) Differences in posture-movement changes induced by repetitive arm motion in healthy and shoulder-injured individuals. Clin Biomech (Bristol, Avon) 26: 123-129.

- Madeleine P, Mathiassen SE, Arendt-Nielsen L (2008) Changes in the degree of motor variability associated with experimental and chronic neck-shoulder pain during a standardised repetitive arm movement. Exp Brain Res 185: 689-698.
- Madeleine P, Voigt M, Mathiassen SE (2008) The size of cycle-to-cycle variability in biomechanical exposure among butchers performing a standardised cutting task. Ergonomics 51: 1078-1095.
- P. Madeleine, T.M.T. Madsen (2009). Changes in the amount and structure of motor variability during a deboning process are associated with work experience and neck– shoulder discomfort. Applied Ergonomics. Volume 40, Issue 5,Pages 887–894.
- Mark Shelhamer, Nonlinear dynamics in physiology: A state space approach, chapter 6: pages 104-109.
- Newell KM, Corcos, Daniel M (1993) Variability and motor control. Champaign: Human Kinetics.
- Nicholas Stergiou, Leslie M. Decker (2011) Human Movement Variability, Nonlinear Dynamics, and Pathology: Is There A Connection?. Hum Mov Sci. 2011 Oct; 30(5): 869–888.
- Nichols PJ, Norman PA, Ennis JR (1979) Wheelchair user's shoulder? Shoulder pain in patients with spinal cord lesions. Scand J Rehabil Med 11: 29-32.
- Pentland WE, Twomey LT (1994) Upper limb function in persons with long term paraplegia and implications for independence: Part I. Paraplegia 32: 211-218.
- Rankin JW, Kwarciak AM, Richter WM, Neptune RR (2012) The influence of wheelchair propulsion technique on upper extremity muscle demand: a simulation study. Clin Biomech (Bristol, Avon) 27: 879-886.
- Rankin JW, Richter WM, Neptune RR (2011) Individual muscle contributions to push and recovery subtasks during wheelchair propulsion. J Biomech 44: 1246-1252.
- Rice IM, Jayaraman C, Hsiao-Wecksler ET and Sosnoff JJ (2013) Relationship Between Shoulder Pain and Kinetic and Temporal-Spatial Variability in Wheelchair Users. Archives of physical medicine and rehabilitation.
- Richman, J.S., Moorman, J.R., 2000. Physiological time-series analysis using approximate entropy and sample entropy. American Journal of Physiology, 278(6), pp. H2039.
- Richman, J.S., Lake, D.E., Moorman, J.R., Sample Entropy. Methods in Enzymology. Academic Press, pp. 172-184.
- Sanderson DJ, Sommer HJ (1985) Kinematic features of wheelchair propulsion. J Biomech 18: 423-429.
- Sethi, A., Davis, S., McGuirk, T., Patterson, T. S., & Richards, L. G. (2013). Effect of intense functional task training upon temporal structure of variability of upper extremity post stroke. Journal of Hand Therapy, 26(2), 132-138.
- Shimada SD, Robertson RN, Bonninger ML, Cooper RA (1998) Kinematic characterization of wheelchair propulsion. J Rehabil Res Dev 35: 210-218.
- Slifkin A. B., Newell K. M. Noise information, transmission and force variability. Journal of Experimental Psychology: Human Perception and Performance. 1999;25:837–851.
- Sosnoff JJ, Newell KM (2006) Are age-related increases in force variability due to decrements in strength? Exp Brain Res 174: 86-94.
- Sosnoff JJ, Newell KM (2006) The generalization of perceptual-motor intra-individual variability in young and old adults. J Gerontol B Psychol Sci Soc Sci 61: P304-310.
- Sosnoff JJ, Rice IM, Hsiao-Wecksler ET, Hsu IMK, Jayaraman C and Moon Y (2015). Variability in wheelchair propulsion: a new window into an old problem. Front. Bioeng. Biotechnol. 3:105. doi: 10.3389/fbioe.2015.00105

- Srinivasan D, Mathiassen SE (2012) Motor variability in occupational health and performance. Clin Biomech (Bristol, Avon) 27: 979-993.
- Srinivasan D, Mathiassen SE (2012) Motor variability in occupational health and performance. Clin Biomech (Bristol, Avon) 27: 979-993.
- Stergiou, N., Buzzi, U., Kurz, M., & Heidel, J. (2004). Nonlinear Tools in Human Movement. 63-90.
- Tochigi, Y., Segal, N.A., Vaseenon, T. And Brown, T.D., 2012. Entropy analysis of triaxial leg acceleration signal waveforms for measurement of decrease of physiological variability in human gait. Journal of Orthopaedic Research, 30(6), pp. 897-904.
- Vieluf S, Temprado JJ, Berton E, Jirsa VK, Sleimen-Malkoun R (2015). Effects of task and age on the magnitude and structure of force fluctuations: insights into underlying neuro-behavioral processes. BMC Neurosci. 2015 Mar 13;16:12. doi: 10.1186/s12868-015-0153-7.
- Vaillancourt D. E., Newell K. M. Changing complexity in human behavior and physiology through aging and disease. Neurobiology of Aging. 2002;23:1–11.
- Sosnoff J. J., Valantine A. D., Newell K. M. Independence between the amount and structure of variability at low force levels. Neuroscience Letters. 2006;392:165–169.
- Sosnoff J. J, Vaillancourt D. E., Newell K. M. Aging and force output performance: The loss of adaptive control of multiple neural oscillators. Journal of Neurophysiology. 2004;91:172–181.
- Yentes, J.M., Hunt, N., Schmid, K.K., Kaipust, J.P., Mcgrath, D. And Stergiou, N., 2013.The appropriate use of approximate entropy and sample entropy with short data sets. Annals of Biomedical Engineering, 41(2), pp. 349-365.

Chapter 6

Prototype and validation of custom wearable technology for manual wheelchair users

Introduction

The reseach studies reported in Chapters 3,4 and 5 all have a common drawback, they are cross section in nature. Further, the data was collected in a laboratory environment. To overcome these limitations a frameowrk to collect wheelchair propulsion data from the user in his/her activity of daily living is proposed. User data collected outside the laboratory is needed to understand the day-to-day propulsion practices that lead to shoulder pain. At present, the widely used equipment for wheelchair propulsion data collection is an instrumented wheel, called, SMARTWheel which is an expensive equipment designed for laboratory based data collection.

Further, research studies to date show that training manual wheelchair users on proper pushing technique and providing continuous feedback on their technique minimizes the risk of shoulder injury. However such technology laden training is not widely accessible to MWCUs and clinicians. This inadequate state of wheelchair propulsion training is due to two factors: (1) the high cost of equipment (~ US \$ 40,000 for force sensing instrumented wheels), and (2) the limited number of available professionals with the capacity to provide such training. Moreover, manual wheelchair users face various barriers (e.g. transportation, cost, etc) that reduce their ability to visit specialized clinics to receive propulsion training. This lack of training exposes them to higher injury risk, and health care costs (i.e. treatment after onset of injury). Consequently, in this concluding chapter of the dissertation the preliminary results from a prototype wearable device, custom developed for manual wheelchair users are reported.

This wearable technology provides individuals access and self-monitor the day-to-day wheelchair propulsion activity through their mobile devices (via wireless integration of wearable sensor data). The goal of this mobile technology is to increase user's awareness on the repetitive usage of their arm and wheelchair propulsion metrics that relate to injury. Integration of such technology and novel research information into rehabilitation and home-based technologies may pave the way for new interventions for tracking, treating, and/or preventing shoulder related overuse pathologies in the manual wheelchair population.

This concluding chapter reports the preliminary validation of this custom developed wearable device. Wheelchair propulsion data obtained from the device will be benchmarked with data from the currently available technologies for tracking manual wheelchair propulsion (SMARTWheel and motion capture). Finally a framework is proposed for incorporating the research finding into the custom developed wearable technology for homebased rehabilitation training purposes.

The custom developed device*

Embedded force and acceleration sensors are placed on a glove. The custom developed device is capable of accurately measuring and securely transmitting real-time wheelchair propulsion data via Bluetooth to any mobile device (Figure 6.1). The sensor/hardware modules are easily attached and detached from most pairs of gloves. The hardware encloses a chargeable lithium ion battery. The data collected can be used to provide a bio-feedback to the user on shoulder activity. For instance, if the arm kinematics during movement is deviating from the usual smooth pattern, the data collected from the device can be used to provide a visual or auditory bio feedback to user to increase their awareness about the situation. The benchmarking of the signal from the custom developed device with SMARTWheel and motion capture will be reported.

^{*}The hardware was designed and developed by Mr.Adam Burns, MS in Civil and Environmental Engineering. I was instrumental in device selection procurement and device testing once the hardware was ready.

Preliminary validations

Force sensor validation

In order to validte the device, the custom device was used on a wheelchair fitted with the SMARTWheel. The participant, pushed the wheelchair wearing the cusotme device on a glove. This enabled us to collect both SMARTWheel data and the data from the sensor simultaneously to benchmark if the force values as recorded from the custom device sensor compare well with that from the SMARTWheels (figure 6.1). The data were recorded at different speeds (figure 6.3 ; 0.5 m/s , 1.1 m/s and 1.3 m/s).

Acceleration sensor validation

To validate the acceleration sensor, the wrist acceleration signatures from the custom device while pushing a wheelchair was compared with wrist acceleration signature data for wheelchair propulsion collected using a 10 camera motion capture system.

Validation results

Our preliminary observation shows that the data obtained using the custom device were comparable with th SMARTWheel and the motion capture device. The peak resultant force at hand-rim compared between the custom device and SMARTWheel were within ~5% magnitude error and the push counts were accurate between both devices (Figure 6.2; Figure 6.3).

In a similar fashion, a comparison of the wrist acceleration signatures for the four propulsion pattern types obtained using the custom device with that a 10 camera motion capture system is shown in figures 6.4 - 6.7. It is to be noted that the plots in these figures are from different individuals, but the features for the wrist acceleration compared well between the two systems.

Proposed framework

The proposed framework is to use this custom developed dvice with cloud integration for continuous monitoring/tracking and training of wheelchair users (Figure 6.8). The goal is to collect propulsion data from the user on a day-to-day basis and provide useful feedback information leading to adopting best practices of wheelchair propulsion that could minimize injury risk.

The information about the three biomarkers identifed in Chapters 3,4 and 5 can be provided as a feedback to keep users aware of their safe propulsion practice. One such sample interface is shown in Figure 6.9. A safe zone of propulsion can be established for each individual and a biofeedback (beep/buzz/email) can be triggered whenever an activity happens to fall above the threshold. For instance, from Figure 6.9, an individuals can be easily observe the alert that around noon and around 5 PM, there were

propulsion activities regarding his/her wheelchair propulsion that could have posed a risk to his/her arms. The user, once made aware of this can review what the activities during these time that could have led to such risk and seek help to make necessary adaptations to reduce risk. As one possible scenario, drawing from Figure 6.9, taking a specific path/route while propelling a wheelchair to go for lunch or evening stroll, may be a reason. The goal is to create this awareness, so that users can self-monitor and reduce situations that could lead to injuries.

List of figures



Figure 6.1 Custom hardware being tested. F- Force sensor. A-Acceleration sensor.



Figure 6.2 Pilot data validation comparing the peak resultant force at palm between the custom device and the SMARTWheel.

Push speed : 0.5 m/s ; Push duration 1 minute

Device	Number of pushes as counted by devices	Mean (SD) Peak force(N)
Custom device	16	37.71(10.13)
SmartWheel	16	35.13(8.86)

Push speed : 1.1 m/s ; Push duration 1 minute

Device	Number of pushes as counted by devices	Mean (SD) Peak force(N)
Custom device	30	53.29(9.67)
SmartWheel	30	50.94(6.32)

Push speed : 1.5 m/s ; Push duration 1 minute

Device	Number of pushes as counted by devices	Mean (SD) Peak force(N)
Custom device	41	67.29(13.90)
SmartWheel	41	64.47(11.89)

Figure 6.3 Comparing peak force at wrist between the SMARTWheel System and prototyped custom developed wearable technology.



Figure 6.4 Wrist acceleration data for a semi-circular propulsion pattern. Acceleration data from custom device (right) identifies all salient features as that from the 10 camera motion capture system (left- Cortex). Sample data shown in this figure are from different participants.



Figure 6.5 Wrist acceleration data for a double-loop propulsion pattern. Acceleration data from custom device (right) identifies all salient features as that from the 10 camera motion capture system (left- Cortex). Sample data shown in this figure are from different participants.



Figure 6.6 Wrist acceleration data for a single-loop propulsion pattern. Acceleration data from custom device (right) identifies all salient features as that from the 10 camera motion capture system (left- Cortex). Sample data shown in this figure are from different participants.



Figure 6.7 Wrist acceleration data for an arc propulsion pattern. Acceleration data from custom device (right) identifies all salient features as that from the 10 camera motion capture system (left- Cortex). Sample data shown in this figure are from different participants.



Figure 6.8 The proposed framework. A cloud integrated system to minimize injury risk in wheelchair users.



Figure 6.9 A sample interface to track arm usage and increase awareness to avoid situation that can cause risk of injury.

Chapter 7

Overall conclusions and future directions

Shoulder pain occurring in manual wheelchair users arising from the mechanical strain of repeated pushing is a multi-faceted problem. To identify biomarkers of shoulder pain in manual wheelchair users, to date research has taken two main approaches, (1) a pure biomechanical approach and (2) improving the wheelchair design (ergonomics). Although valuable in establishing clinical guidelines on best practices of wheelchair propulsion, the previous research have had very limited success in extracting biomarkers of shoulder pain from propulsion data. Consequently, this dissertation started with the main aim to identify biomarkers of shoulder pain in manual wheelchair users using a multi-disciplinary approach.

The dissertation implemented theories and approaches from occupation ergonomics, human motor control and non-linear dynamics to identify three such biomarkers. Chapter 3 of this dissertation examined the jerk during recovery phase of wheelchair propulsion to investigate the differences in kinematics between manual wheelchair users with and without shoulder pain. Chapter 4 of this dissertation explored the <u>trunk kinematics</u> during wheelchair propulsion between groups with and without shoulder pain. Chapter 5 investigated the <u>time dependent structure in variability of</u> wheelchair propulsion variables between groups with and without shoulder

pain. Finally Chapter 6, prototyped and benchmarked the data quality from a custom developed wearable device, which has capabilities to implement the findings from Chapters3,4 and 5 in real-time. Further, Chapter 6 also proposed a framework to implement this prototyped wearable device for continuous propulsion data monitoring via a mobile device and home based rehabilitation training.

Major findings and implications

Shoulder pain and Jerk in wheelchair propulsion

Jerk, the third derivative of position has been widely employed in clinical rehabilitation and human motor control research to quantify movement smoothness and evaluate the performance of upper limb tasks (Hogan et al., 1987, Flash., 1990, Chang et al., 2005, Caimmi et al., 2008). Occupational ergonomics research has revealed distinct differences in arm jerk between movements in individuals with and without shoulder pain (Cote et al., 2005). However, there has been minimal inquiry on understanding wheelchair propulsion kinematics from a human movement ergonomics perspective. Consequently, this investigation employed an ergonomic metric, jerk, to characterize the recovery phase kinematics of two recommended manual wheelchair propulsion patterns: semi-circular and the double loop. Further it examined if jerk is related to shoulder pain in manual wheelchair

users. Two hypotheses were postulated, namely, (H1) that individuals using a SC recovery pattern will experience lower jerk magnitudes at their wrists than individuals using a DLOP recovery pattern. (H2) that individuals with shoulder pain will minimize peak jerk magnitude at their upper arm joints during the recovery phase kinematics in an effort to avoid pain. H1 rested on the logical rationale that the arm's movement trajectory during a SC pattern is simpler than a DLOP pattern. H2 was based on the observation that the neuromuscular system avoids large acceleration changes to avoid pain (Berret et al., 2008).

Overall observations were, (1) the recovery phase kinematics of individuals using a SC recovery pattern placed lower jerk magnitudes than those using a DLOP and (2) mWCUs with shoulder pain had lower peak jerk magnitude during the recovery phase of wheelchair propulsion.

The logical reason for the DLOP recovery pattern to incur greater J_c was attributed to the joints kinematics undergoing sharp directional turns, leading to frequent switching between acceleration and deceleration during the recovery trajectory. In contrast, when executing a SC pattern the arm underwent less directional change leading to relatively lower jerk. With context to shoulder pain and jerk, it was maintained that individuals with shoulder pain adopt a smoother arm motion pattern to reduce momentary discomfort at the shoulder during wheelchair propulsion.

In conclusion the results from the Chapter 3 led to the following generalized conclusions:

- adopting jerk based quantification of wheelchair propulsionkinematics is worthwhile and yields insightful inferences;
- (2) investigating the recovery phase kinematics in the contextof shoulder pain is as important as studying the push phase,and
- (3) in the future it may be beneficial to incorporate jerk based metric into rehabilitation practice.

Shoulder pain and trunk kinematics in wheelchair propulsion

Trunk kinematics during wheelchair propulsion is known to influence propulsion biomechanics and also bears implication for shoulder injury. The main aim of this investigation was to study the trunk kinematic differences between manual wheelchair users with and without shoulder pain. Research guidelines from maintain that propelling a manual wheelchair with a 'trunk flexed / anterior-tilt' position exposes mWCUs to secondary injury risk (Rodgers et al., 2000, Rodgers et al., 2001, Rankin et al., 2011, Gagnon et al., 2015, Gagnon et al., 2009, Sanderson DJ et al., 1985, Vanlandewijck YC et al., 1989, Chow et al., 2009, Yang et al., 2006, Rice et al., 2004). However, presently there is very limited information regarding trunk kinematics differences between groups propelling a manual wheelchair with and without shoulder pain. Consequently, this cross-sectional study analyzed the trunk kinematics during manual wheelchair propulsion in a group of experienced adult MWCUs with and without shoulder pain. We hypothesized that the trunk kinematics of individuals propelling a manual wheelchair with shoulder pain will significantly differ from the trunk kinematics of the group without shoulder pain. The net drift in the trunk position (during stead state propulsion (~135 cycles)) from the initial reference position (rest) along the sagittal plane (X direction) was computed and the group means statistically tested to validated the postulated hypothesis.

Our investigation revealed that individuals with shoulder pain had larger net deviation trunk position in sagittal plane (trunk movement in sagittal plane) from the initial reference position than those without shoulder pain. This observed phenomenon was potentially attributed to be a compensatory mechanism to minimize discomfort to shoulder during propulsion. This attribution is consistent with recommendations from occupation ergonomics literature (Madeleine et al., 2008, Lomond et al., 2010, Lomond et al., 2011). Individuals with neck/shoulder pain performing rhythmic repetitive occupational tasks adopted a spatial strategy minimizing the movement of their trunk to minimize discomfort (Madeleine et al., 2008, Lomond et al., 2010, Lomond et al., 2011). Further, it was discussed that from a pure human movement ergonomics perspective, when the trunk was flexed forward, the arm joints grab the hand-rim with greater internal

rotation at the initiation of the start of the proceeding push phase. It is well known that application of load to arm joints with at such orientation contributes to increased risk of causing injury to joints (Bridger RS 2009).

Based on these rationale's and observed results overall it was concluded that studying trunk kinematics is related to shoulder pain and incorporating trunk kinematics based measures, may provide additional knowledge to understand the adaptive strategies employed by mWCUs with shoulder pain.

In conclusion the results from the Chapter 4 led to the following generalized conclusions:

- (1) trunk kinematics can be used to differentiate betweenmanual wheelchair users with and without shoulder pain;
- investigating the trunk kinematics over the entire propulsion
 cycle (push + recovery) captures the adaptive strategy as
 opposed to only studying the push phase, and
- (3) including data from the entire trials (i.e. in our case the trial has ~135 cycles on average) for analysis may have facilitated the capture of adaptive dynamics, better than merely using a portion or a small fraction of data from the full trial (i.e. for example only, 10 to 20 cycles from a trial rather than the full trial data).

Shoulder pain and time dependent structure in propulsion variables

Based on the tenets of the *loss of complexity* hypothesis of aging (Lipsitz et al., 2004) the time-dependent structure of motor output has been associated with physiological complexity and shown to provide novel information concerning the health of the musculoskeletal system (Hausdroff et al., 1997, Sosnoff et al., 2006, Stregiou et al., 2011). Specifically, it has been proposed that musculoskeletal injury leads to motor fluctuations that are more structured (i.e. more regular – loss of complexity). The amount of variability is usually quantified using the distribution statistics (standard deviation, coefficient of variation). However, dynamical measures of motor variability have been found to be more sensitive to pathology than measures involving distributional statistics (Slifkin et al., 1999, Stergiou et al., 2011, Stergiou et al., 2004).

The size and time dependent structure of motor variability have been shown to provide unique information concerning the control and health of the neurophysiologic system (Lipsitz et al., 2004, Sosnoff et al., 2006, Hausdroff et al., 1997, Newell et al., 1993, Madeleine et al., 2009). Recent research indicates that the size of variability in wheelchair propulsion mechanics is related to shoulder pain (Sosnoff et al., 2015). This association between size of motor variability and pain is consistent with observations from occupation biomechanics and human motor control literature (Srinivasan et al., 2012, Madeleine et al., 2009, Hamill et al., 2012 Stergiou
et al., 2011).

Presently there is no information relating shoulder pain and variable structure in wheelchair propulsion. Research evidence from occupation ergonomics shows that the structure of variability is sensitive discomfort level arising from repetitive strain injuries (Madeleine et al., 2009). However to date, the structure in variability of wheelchair propulsion variables have not been investigated. Consequently chapter 5 investigates the *loss of complexity* hypothesis. It was hypothesized that manual wheelchair users with shoulder pain will demonstrate lower complexity compared to those without shoulder pain. Consequently the purpose of this investigation is to examine if shoulder pain and the variable structure in wheelchair propulsion are related.

In this investigation, sample entropy (SampEn), a widely utilized approach to quantify the structure of variability was used (Yentes et al., 2013, Tochigi et al., 2012). The SampEn for the cycle-to-cycle fluctuations in contact angle and inter push time interval was computed. As a first step a simple surrogate analyses was performed to establish that the structure found in these variables were not random and had some meaningful structure. Following this the group mean values of SampEn were compared between the groups with and without shoulder pain to validate the acceptance or rejection of the postulated hypotheses.

Our results revealed that the SampEn for contact angle was able to differentiate between groups with and without shoulder pain. This in agreement with occupation ergonomics research (Madeleine et al., 2009). The higher SampEn magnitude observed for the contact angle at the handrim in mWCUs with shoulder pain could be a manifestation of their compensatory strategies to minimize shoulder discomfort when performing the repetitive propulsion task. Consistent with previous research where higher discomfort (pain) levels were associated with higher spatial complexity (Madeleine et al., 2009), SampEn of contact angle was positively correlated with the self-reported pain scores.

The SampEn for time interval was not able to discriminate between groups with and without shoulder pain. A possible reason for this could be the disruption in temporal structure of the time interval due to the visual feedback on propulsion speed (Hausdroff et al., 1996).

Overall in conclusion, this investigation evinced three novel observations:

(1) variability observed in the fluctuation in contract angle and time interval during manual wheelchair propulsion is structured;

(2) individuals with shoulder pain exhibited higher SampEn magnitude during wheelchair propulsion compared to those without pain, and

(3) SampEn measure correlated significantly with the amount of self-

reported shoulder pain.

This observation is consistent with reports from occupational ergonomics and motor control research that variability of motor output may offer meaningful insights concerning health and function (Madeleine et al., 2008, Madeleine et al., 2009, Vaillancourt et al.,2002, Sosnoff et al., 2015, Vieluf et al., 2015, Srinivasan et al., 2015). Overall in conclusion, this investigation evinced that incorporating non-linear dynamics based measures in wheelchair propulsion analyses may provide new knowledge and can be an important tool for better health, diagnosis, and therapeutic interventions to prevent shoulder pain pathology in mWCUs.

Limitations

Since all the studies were from the same data set, the major limitations are common for all the three studies (Chapters 3, 4 and 5). A major limitation of this dissertation is the cross sectional nature of the data. The data were collected in a laboratory setting with roller dynamometer setup, so it is not clear if these results would occur in real life propulsion environment. The injury demographics of our sample was diverse. Consequently, it is possible that the between group differences observed for the biomarkers were due to different disability demographics. Another limitation is that, the pain scores were self-reported and no radiographic or ultrasonic information on pain were collected. Though significant, these

demographic and environmental limitations are relatively common to wheelchair propulsion research. Finally, despite these limitations, this investigation provides novel contributions that are important and compliments the research findings from previous research (Sosnoff et al., 2015, Madeleine at al., 2009, Srinivasan et al., 2012, Madeleine et al., 2009). All these findings should be repeated in larger samples to check repeatability of result trends.

Future directions

In this section, first, some future research directions/improvements for each of the studies (Chapters 3, 4, and 5) will be proposed. Following this, in conclusion, a future research framework to address the overall limitation identified is being proposed.

Shoulder pain and jerk during wheelchair propulsion.

- Improvement to method: One of the methodological limitation faced in the jerk analysis was the constraint to group participants by pattern type. To overcome this limitation a dimensionless version of jerk metric is proposed. Such dimensionless approach will enable comparison collapsing across recovery pattern types.
- <u>Addressing open research question in wheelchair propulsion</u> <u>literature</u>: A main open question that still lies unanswered in wheelchair propulsion literature is, how does the human motor system

choose a particular type of recovery pattern type given the environment, physical state and task constraints?

One way to seek answer to this open question is to verify if individuals choose propulsion pattern based on cost criteria which is encompasses a jerk minimization criteria. For this a pilot study with able bodied individuals is proposed. Individuals without any prior experience with manual wheelchair can be trained to use all the four propulsion pattern and on reaching certain level of propulsion skill, can be asked to propel with various task constraints (ramp, different floor types, different speed etc) and the self chosen pattern kinematics for these conditions can be analyzed to see if individuals minimizes jerk to choose the pattern.

Shoulder pain and jerk during wheelchair propulsion.

 Improvement to method: As a methodological improvement, a study of the acceleration profile of the trunk is proposed. Research evidence from occupational ergonomics, show that the trunk acceleration metrics and adaptive responses to neck/shoulder pain are related (Madeleine et al., 2008). On the same lines, a jerk based analysis of trunk kinematics is also warranted.

• Addressing open research question in wheelchair propulsion

<u>literature</u>: One unexplored research question in wheelchair propulsion literature is, how does constraining or un-constraining the trunk affect wheelchair propulsion biomechanics/performance?

To answer this, a study with able bodied novice individuals is proposed. The participants can be trained to propel wheelchair in one specific/different pattern. Once they reach a certain skill level, the participants can propel the wheelchair on a treadmill with different task constraints (speed and different ramp), (1) trunk unconstrained, (2) trunk partially constrained to backrest and (3) trunk fully constrained to backrest. Results from such study will provide novel knowledge to understand what strategy the neuromuscular system adapts given the task constraint.

Shoulder pain and structure in variability during wheelchair propulsion.

 Improvement to method: One main limitation of the methodology for the SampEn investigation was the small data record length. Our work is the first to apply a non-linear dynamics approach to wheelchair propulsion. There is no wheelchair propulsion data specific information on guidelines to choose the sample entropy parameters (embedding dimension (m), tolerance (r)). A proposed improvement to the chapter 5 in terms of methodology will be to collect wheelchair propulsion data with for longer duration (6 to 8 minutes) to be able to study the influence of m and r on wheelchair propulsion data. Longer data length will also enable complimenting the SampEn results with other non-linear dynamic methods like Lyapunov exponent (LyE) and de-trended fluctuation analysis (DFA). In general, adopting a combination of non-linear methods is recommended as best practice (Stergiou 2004). A second methodological improvement will be to remove the visual feedback for speed to study the time dependent structure in the time interval. In the current procedure in Chapter 5, the visual feedback may have disrupted the temporal structure for the time interval variable.

• Addressing open research question in wheelchair propulsion <u>literature</u>: It is an open question to study the relationship between

SampEn and propulsion variables at the shoulder. Presently, the Chapter 5 only evaluates, the SampEn for propulsion variables at hand-rim. It is not clear if the same trend in SampEn will exist at the site of pain (shoulder). This is an open area for future enquiry.

A wearable technology for longitudinal data collection

As was stated in the limitation section, one of the main limitations with wheelchair propulsion research in general is the very limited access to longitudinal data and propulsion data outside lab environment. The currently available instrument wheels (SMARTWheel and OptiPush) are very expensive/ bulky and not viable options for individualized home based longitudinal data collection. To overcome this limitation the dissertation proposes a 3 to 5 year longitudinal study tracking novice manual wheelchair users longitudinally using an improved version of the wearable technology prototyped in chapter 6. The wearable technology can also be used to continuously track the identified biomarkers to validate if these variables can track/aid in preventing the progression of shoulder pain n manual wheelchair users.

Chapter 8

Intellectual contributions to wheelchair propulsion research

To study shoulder pain in manual wheelchair users, the traditional approach for the past two decades focused only on the mean wheelchair propulsion parameters (biomechanics approach) and improvements to wheelchair build (wheelchair ergonomics). Given that the association between shoulder pain and manual wheelchair propulsion is multi-faceted in nature, a multi-disciplinary approach to analyze and address this problem is more appropriate. Such multi-disciplinary approaches can provide better understanding of the pathology and may lead to new knowledge for better monitoring/tracking/ and prevention of shoulder injury in manual wheelchair users.

This dissertation integrated approaches and theories from human motor control, occupation ergonomics and non-linear synamics to investigate shoulder pain in manual wheelchair users. The dissertation research was successful in identifying biomarkers that relate to shoulder pain using the multi-disciplinary approach proposed. Futher, the dissertation also prototype tested a custom developed wearable technology that has the potentital to translate the research findings from the dissertation for the benefit of manual wheelchair users.

This dissertation made the following intellectual contributions to wheelchair propulsion research:

- identified three novel metrics that relate shoulder pain and manual wheelchair propulsion;
- <u>opened new avenues for future research</u> in wheelchair propulsion research;
- proposed methods using a multi-disciplinary approach that has <u>potential to addreess open questions</u> in wheelchair propulsion literature;
- established that <u>investiagating the recovery phase during</u>
 <u>wheelchair propulsion</u> yields novel and useful inference to understand adaptive stragtegies during wheelchair propulsion;
- validated the prototype of an affordable custom developed <u>wearable device</u> with the potential to translate the research findings to acual wheelchair users.